MODELING OF EXISTING GOVERNMENT BUILDINGS FOR ECBC 2018-R COMPLIANCE

A thesis submitted to the

UPES

For the Award of *Doctor of Philosophy*

in

Energy Engineering

By

Rakesh Dalal SAP ID: 500065478 August 2024

Supervisor

Dr. Devender Kumar Saini Senior Associate Professor Electrical Cluster, UPES



School of Advanced Engineering UPES Dehradun – 248007: Uttarakhand



DECLARATION

I declare that this thesis submission titled 'MODELING OF EXISTING GOVERNMENT BUILDING FOR ECBC 2018-R COMPLIANCE' has been prepared by me under the guidance of Devender Kumar Saini Associate Professor, Department of Electrical Engineering UPES. No part of this thesis has formed the basis for the award of any degree or fellowship previously.



Rakesh Dalal SAP ID 500065478, Doctoral Research Fellow, School of Engineering, UPES, Dehradun-248007 Date: 21 August 2024





CERTIFICATE

1 certify that Rakesh Datal has prepared his thesis entitled "MODELING OF EXISTING GOVERNMENT BUILDING FOR ECBC 2018-R COMPLIANCE", for the award of PhD degree at the UPES, under my guidance. He has carried out the work at the School of Advanced Engineering, UPES.

ds

Dr. Devender² Kumar Saini Sr. Associate Professor, Electrical Cluster School of Advanced Engineering, UPES Dehradun -248007, Uttarakhand Date: August 2024

Energy Acess: Bidholi Via Piere Nagar, Dehradud 48 007 (Utarakhand), India T. +91135277013. 32756253:54/9., 3776201,9997398474 F. +911352776090/95 Knowledge Acess: Kandoli Via Piere Nagar, Dehradus - 240 007 (Utarakhand), India T. +918171979023/2/3, 7060111775

ADVANCED ENGINEERING | COMPUTER SCIENCE | DESIGN | BUSINESS | LAW | HEALTH SCIENCES AND TECHNOLOGY | LIBERAL STUDIES

ABSTRACT

Techno-economic viability of rooftop solar PV systems for Indian residential buildings was undertaken in this study, 34 major cities including 15 state capitals were considered for the RET screen simulation. In addition to location, utility grid tariff, system cost, and other economic parameters were plugged into RETscreen simulation software for ascertaining the viability of 3 kWp and 2 kWp rooftops solar PV with and without subsidy. The study concluded that in the absence of subsidy, 2kWp rooftop solar PV could not achieve grid parity in any Indian state. However, in the case of 3 kWp rooftops solar PV, 5 states could match the grid tariff even in the absence of state subsidy and 7 states could achieve grid parity with government subsidy. Yet, most Indian states could not achieve grid parity for residential rooftop solar PV.

The Energy Performance Index (EPI) is an indicator by the Bureau of Energy Efficiency (BEE) for awarding energy stars to the building. The study discovers a large EPI gap exists for residential buildings to attain a five-star energy label. Based on the EPI gap, residential energy tariffs, the energy consumption, energy a grid-integrated rooftop solar Photo Voltaic (PV) system was considered for bridging this EPI gap for the higher energy star label.

iv

A case study of a residential building in Delhi was undertaken to explore the potential of grid-connected rooftops solar photovoltaic. RETScreen simulation software was used to assess the techno-economic potential of a grid-integrated, 3 kWp rooftops solar PV plant. The study establishes that by using a gridintegrated rooftop solar PV an additional two energy stars can be achieved by existing buildings. 3 kWp rooftop solar PV economics with a subsidy is a viable retrofit and invested amount can be recovered within 3 to 7 years in Indian cities. The Energy Star label to the residential building is awarded against the notified standard by the regulatory body and Electric Vehicle (EVs) has not been catered to as a plugged load for the residential buildings. Energy consumption of an existing residential building is taken from a study already carried out and is compared with the requirement of the Indian residential star labeling program with EV as a plugged load. An annual energy gap of 6060 kWh for the existing residential buildings under consideration in this study for five-star building energy labels increased to 7784 kWh if the EV load is added to the building load. The residential building will lose two energy stars if it caters to EV load and for bridging this energy gap, replacement of existing electrical appliances with five-star rated energy appliances, employing grid-connected rooftop solar PV, and retrofitting of the building envelope are considered. The techno-economic potential of rooftop solar PV and building envelope retrofitments for existing residential buildings are explored using RETScreen and e-QUEST software respectively. The study establishes that the installation of rooftop solar PV can cater to additional EV load and can bridge half and three-fourths of the energy gaps for achieving five energy stars for an existing building with EV and without EV respectively and it is the most economical among the options explored in this study. Target Energy Performance Index (EPI) is achievable by high-end energy consumers (12000 kWh/year) by additional measures of the replacement of inefficient electrical appliances and building envelope retro fitment in addition to the installation of rooftop solar PV.

ACKNOWLEDGEMENT

I take this opportunity to express my sincere thanks to my supervisor Dr. Devender Kumar Saini for the guidance and support. I would also like to express my sincere gratitude towards my early-stage co-supervisor Dr. Kamal Bansal who along with Dr. Madhu Sharma helped me with topic selection as well as in setting the research objectives. It was a privilege and a great experience working under the guidance of all the above supervisors.

I am also grateful to the Indian Air Force for allowing me to undertake this research. Further, I am very thankful to the UPES for providing me the opportunity of pursuing a Ph.D. in a very cordial environment.

In the end, I would like to acknowledge the contribution of my wife, children, and parents, without their support and sacrifice this work would not have been possible.



Rakesh Dalal SAP ID 500065478, Doctoral Research Fellow, School of Engineering, UPES Dehradun-248007

LIST OF CONTENTS

| DECLARATIONii |
|--|
| ABSTRACTiv |
| ACKNOWLEDGEMENTvii |
| LIST OF FIGURESxi |
| LIST OF TABLESxiii |
| Nomenclaturexv |
| CHAPTER 1_INTRODUCTION1 |
| 1.1 BRIEF BACKGROUND1 |
| 1.2 SCOPE & FOCUS OF THE STUDY4 |
| 1.3 RELEVANCE OF THE STUDY5 |
| 1.4 OBJECTIVE OF THE STUDY5 |
| 1.5 OVERVIEW OF STRUCTURE |
| CHAPTER 2_LITERATURE SURVEY & CASE STUDY8 |
| 2.0 PROLOGUE |
| 2.1 SCOPE OF LITERATURE SURVEY8 |
| 2.2 ESTABLISHING BUILDING ENERGY CODES - A GLOBAL PERSPECTIVE14 |
| 2.3 INDIA'S INITIATIVE FOR ESTABLISHING BUILDING12 |
| 2.4 BUILDING INTEGRATED ROOFTOP SOLAR PV17 |
| 2.5 MODELLING FOR BUILDING ENERGY CODE COMPLIANCE BY SIMULATION |
| 2.6 EPILOGUE |

| CHAPTER 3_RESEARCH DESIGN | 23 |
|--|-------------|
| 3.0 PROLOGUE | 23 |
| 3.1RESEARCH GAPS | 23 |
| 3.2 MOTIVATING FACTOR | 25 |
| 3.3 OBJECTIVES | 25 |
| 3.4 METHODOLOGY AND DATA COLLECTION | 26 |
| 3.5 EPILOGUE | 28 |
| CHAPTER 4_STATUS OF ROOFTOP SOLAR PV GRID PARITY RESIDENTIAL BUILDINGS IN INDIA | FOR |
| 4.0 PROLOGUE | 29 |
| 4.1 GRID PARITY CONCEPT | 29 |
| 4.2 COST OF ENERGY FOR RESIDENTIAL CONSUMERS IN INDIAN STAT | ES31 |
| 4.3 EVALUATION OF ENERGY GENERATION FROM ROOFTOP SOLAR P INDIAN CITIES | 'V IN 33 |
| 44 ECONOMIC EVALUATION OF ROOFTOP SOLAR PV PLANT | |
| 4.5 EPILOGUE | 50 |
| CHAPTER 5_STUDY OF RESIDENTIAL BUILDINGS FOR_ENE STARS LABELING | CRGY |
| 5.0 PROLOGUE | 51 |
| 5.1 RESIDENTIAL BUILDINGS STAR LABEL | 51 |
| 5.2 CASE STUDY FOR QUANTIFYING THE ENERGY CONSUMPTION OF A RESIDENTIAL BUILDING | A 52 |
| 5.3 BENCHMARKING OF BUILDING ENERGY CONSUMPTION | 54 |
| 5.4 EVALUATION OF ENERGY GENERATION BY ROOFTOP SOLAR | 57 |
| 5.5 ECONOMIC EVALUATION OF ROOFTOP SOLAR PV | 59 |

| CHAPTER 6 IMPACT OF ELECTRIC VEHICLES ON ENERGY |
|---|
| STARS FOR EXISTING RESIDENTIAL BUILDINGS IN INDIA AND |
| ITS MITIGATION |
| |
| 6.0 PROLOGUE |
| 6.1 EV SCENARIO IN INDIA |
| 6.2 EVALUATING EV LOAD ON BUILDING |
| 6.3 TECHNO-ECONOMIC ASSESSMENT OF ENERGY STAR APPLIANCES FOR RESIDENTIAL BUILDINGS |
| 6.4 TECHNO-ECONOMIC ASSESSMENT OF ENERGY SAVING BY ROOFTOP SOLAR PV |
| 6.5 TECHO-ECONOMIC ASSESSMENT OF THE BUILDING ENVELOPE RETROFITMENT |
| 6.6 EPILOGUE |
| CHAPTER 7_CONCLUSION AND RECOMMENDATIONS FOR FUTURE WORK |
| 7.1 CONCLUSION |
| 7.2 FUTURE WORK |
| REFERENCES |
| APPENDIX – 'A'_BIODATAND RESEARCH PAPERS PUBLISHED |
| APPENDIX – 'B' PLAGRISM REPORT95 |

LIST OF FIGURES

| Figure 1.1 Energy consumption by the residential sector in different countries | 2 |
|--|-----------------|
| Figure 1.2 Electricity consumption by different sectors in India (2020-21) | 3 |
| Figure 1.3 Increase in electricity consumption by different sectors (1990-2019) | 3 |
| Figure 2.1 Availability of building code for new buildings across the globe [18] | 10 |
| Figure 2.2 Availability of building code for existing buildings across the globe [18] | 11 |
| Figure 2.3 India's initiative in energy conservation [25] | 13 |
| Figure 2.4 Global solar radiation map for Indian states | 18 |
| Figure 3.1 Research Methodology for rooftop solar PV | 16 |
| Figure3.2 Flow chart for the study of EV scenario | 27 |
| Figure 4.1 Status of grid-connected solar PV in India [76] | 30 |
| Figure 4.2 Installation status of rooftop solar PV in the different sectors [77] | 30 |
| Figure 4.3 Grid parity status of 3 kWp rooftops solar PV in Indian cities (including | |
| subsidy) | 34 |
| Figure 4.4 Grid parity status of 3 kWp rooftops solar PV in Indian cities (in the abso | ence |
| of subsidy) | 35 |
| Figure 4.5 Grid parity status of 2kWp rooftop solar PV in Indian cities (including | |
| subsidy) | 36 |
| Figure 4.6 Grid parity status of 2kWp rooftop solar PV in Indian cities (in the absen | ce of |
| subsidy) | 37 |
| Figure 4.7 Payback period of 3 kWp rooftops solar PV in Indian cities | 40 |
| Figure 4.8 Payback period of 2 kWp rooftops solar PV | 41 |
| Figure 4.9 IRR of 3 kWP rooftop solar PV | 42 |
| Figure 4.10 IRR of 2kWp rooftop solar PV | 43 |
| Figure 4.11 3 kWp Rooftop solar PV grid parity with subsidy | 46 |
| Figure 4.12 3 kWp Rooftop solar PV grid parity without subsidy | 47 |
| Figure 4.13 2 kWp Rooftop solar PV grid parity with subsidy | 48 |
| Figure 5.1 Layout of flats as the RETscreen | 52 |
| Figure 5.2 Energy consumption of residential consumers | 53 |
| Figure 5.3 Categories of consumers based on annual electricity consumption (2017- | l 8) 55 |
| Figure 5.4 Categories of consumers based on annual electricity consumption (2018- | L9) 56 |
| Figure 5.5 Solar radiation at the study site as per RETscreen | 57 |
| Figure 5.6 Energy star building status 2017-18 post 3 kWp rooftops solar PV | 59 |
| Figure 5.7 Energy star building status 2018-19 post 3 kWp rooftops solar PV | 59 |
| Figure 6.1 4-wheeler EV sales in India (2017-18 and 2023-24) | 63 |
| Figure 6.2 Probability distribution of arrival and departure of EV | 65 |
| Figure 6.3 Cumulative probability distribution of arrival/departure of EV | 66 |
| Figure 6.4 PDF of daily distance traveled by EV | 67 |

| Figure 6.5 Building Model Simulated by eQUEST | 75 |
|--|----|
| Figure 6.6 Distribution of the building's annual energy load | 75 |

LIST OF TABLES

| Table 1.1 Consumption of Primary energy by different country groups | 1 |
|---|----|
| Table 4.1 Residential energy consumer groups | 31 |
| Table 4.2 Energy charges for residential consumers [79] | 32 |
| Table 4.3 TATA solar rooftop solar PV system cost [79] | 33 |
| Table 4.4 Parameters plugged in the RETsceen module. | 39 |
| Table 4.5 Parameters for Grid parity for rooftop solar PV for Indian states | 46 |
| Table 5.1 Star labeling requirement [EPI(X)] in different climate zones [82] | 52 |
| Table 5.2 Building Location details | 53 |
| Table 5.3 Energy tariff for Delhi residential buildings[83] | 55 |
| Table 5.4 Residential consumer classification based on annual energy consumption. | 55 |
| Table 5.5 Energy generation by rooftop- solar calculator [84] | 58 |
| Table 5.6 Rooftop solar PV energy simulation using RETscreen | 59 |
| Table 5.7 NPV and Payback period of 3kW rooftop solar PV | 61 |
| Table 5.8 Energy tariff and solar radiation of Indian cities | 61 |
| Table 5.9 Techno-economic results of 3kWp rooftop solar PV | 62 |
| Table 6.1 Consumer categorization based on annual energy consumption. | 69 |
| Table 6.2 Electricity charges for Delhi residential building [80] | 70 |
| Table 6.3 Residential Electric Appliances Operating Parameters | 71 |
| Table 6.4 Energy saving by replacement of residential building appliances. | 72 |
| Table 6.5 Economics of replacement of appliances | 73 |
| Table 6.6 Rooftop Solar Installation [52] | 74 |
| Table 6.7 Residential rooftop solar PV economics | 74 |
| Table 6.8 Energy conservation by adding insulation to the building envelope | 77 |
| Table 6.9 Economics of Building Envelope Retrofit | 77 |

LIST OF APPENDIXES

| APPENDIX-A | BIO | DATA | AND | RESEARCH | WORK | PUBLISHED |
|-------------|-----|--------|--------|---------------------------------------|------|-----------|
| | | | | | | 93 |
| APPENDIX -B | PLA | GARISM | REPORT | · · · · · · · · · · · · · · · · · · · | | 94 |

NOMENCLATURE

Acronyms

- BEE Bureau of Energy Efficiency
- CAGR Compound Annual Growth Rate
- CCE Cost of Conserved Energy
- CEA Central Electricity Authority
- CO₂ Carbon dioxide
- ECBC Energy Conservation Building Code
- EER Energy Efficiency Ratio
- EIA Energy Information Administration
- EPI Energy Performance Indicator
- EV Electric Vehicle
- FAME Faster Adoption and Manufacturing of Electric Vehicles
- GW Giga Watt
- IEA International Energy Agency
- ISEER Indian Seasonal Energy Efficiency Ratio
- kWh Kilo Watt Hour
- kW_p Kilo Watt peak
- kVA Kilo Volt Ampere
- MMRE Ministry of New and Renewable Energy
- Mtoe Million tons of oil equivalent

- MW Mega Watt
- PAT Perform Achieve and Trade
- PP Payback Period
- PPP Purchasing power Parity
- NPV Net Present Value
- SPV Solar Photovoltaic
- TWh TerraWatt Hour
- TPES Total Primary Energy Supply

CHAPTER 1

INTRODUCTION

1.1 BRIEF BACKGROUND

The current trends of energy shortage and awareness of many environmental (GHG emissions) related issues have raised concerns worldwide about energy consumption and related energy efficiencies. As per the British Petroleum (BP) report the global primary energy demand has increased more in non-OECD countries as compared to OECD and EU countries in the period 2007-17. The consumption of primary energy by the country groups is tabulated below[1].

| Country Group | (Primary energy-Mtoe) | (Primary energy-Mtoe) |
|---------------|-----------------------|-----------------------|
| | Yr 2007 | Yr 2017 |
| OECD | 5693.9 | 5605.5 |
| Non-OECD | 5894.5 | 7906.1 |
| EU | 1823.9 | 1689.2 |
| Total | 11588.4 | 13511.2 |

Table 1.1 Consumption of Primary energy by different country groups

As per a study, global oil consumption will increase by around 30% from 2007 to 2035, while other fossil fuels (coal and natural gas) consumption will grow by 50%. As per the IEA in business as usual scenario, the emissions of carbon

dioxide (CO2) associated with energy-related activities will increase by a hundred percent by 2050 [2]. The building sector alone contributes more than 40% of the total energy consumption. The CO₂ emission from the residential sector of the top ten countries/groups is shown in Figure 1.1[3].



Figure 1.1 Energy consumption by the residential sector in different countries The consumption status of different sectors of India is represented in Figure 1.2. As per IEA data, India's residential sector is the leading consumer of electricity and the consumption has nearly grown a hundred percent between 2010 and 2019 growth of different sectors in this period is represented in Figure 1.3 [3].



Figure 1.2 Electricity consumption by different sectors in India (2020-21)



Figure 1.3 Increase in electricity consumption by different sectors (1990-2019) Standard and Labeling (S&L) is a policy instrument adopted by countries to increase awareness of the energy performance of appliances equipment and buildings among consumers. S&L aims to channel consumers' purchases towards energy-efficient products by providing Energy Star information on the products. This is also extended to the building sector where building components have energy labeled so that building energy consumption matches the approved design.

India promulgated the Energy Conservation Building Code (ECBC) in 2007 to improve the energy efficiency of commercial buildings. Initially, this code was applied to all buildings with plugged electrical loads of more than 500 kW /contact demand of more than 600 kVA. In 2017, the code was modified for building/building complexes with an electrical load of more than 100 kW/ contract demand of 120 kVA were brought into the ambit of ECBC. Energy code provides design guidelines for improving building envelope for better energy efficiency. In 2018, a residential buildings energy code was introduced and it applies to all the residential buildings that are built on plot areas greater than or equal to 500 m² [4]. Star labeling program for residential buildings was also introduced in India to improve energy efficiency in the residential buildings sector [5]. For applying Energy Star for a residential building, there is no lower limit on the building built-up area or minimum connected electric load. The Energy Performance Indicator (EPI) is taken as an indicator for awarding the star label where building annual electricity consumption/ m^2 is compared with benchmarked EPI promulgated by the Bureau of Energy Efficiency (BEE).

1.2 SCOPE & FOCUS OF THE STUDY

Although some guidelines are available for reducing the energy consumption of a residential building there is no holistic approach to address the potential of energy saving through active and passive measures in the Indian context for achieving the desired energy star label. Techno-economic analysis of the measures is to be gauged so that the Energy Star label for residential buildings is achieved through effective cost optimization. Installation of rooftop solar PV has been impetus by the Indian government and a flat 30% subsidy is extended for installation of rooftop solar PV up to 3 kW. Grid parity of rooftops which has not been explored by any of the studies in the Indian context can also be examined for achieving higher energy stars.

1.3 RELEVANCE OF THE STUDY

The study aims to address a live problem that is being faced by the government's existing residential buildings in attaining the energy star as proposed by the Bureau of Energy Efficiency (BEE). The cost associated with the retrofitting of the building needs optimization so that the ownership cost of the building Energy Star is minimised.

1.4 OBJECTIVE OF THE STUDY

- ✓ To propose a cost-effective solution for attaining the desired energy stars by residential buildings in hot and dry climates.
- Analysing the impact of EVs on building energy Stars and exploring the mitigation of this scenario, as an Electric Vehicle (EV) is not considered a plugged load for the Building Star label program.

1.5 OVERVIEW OF STRUCTURE

The structure of the thesis progresses from identifying problems and culminates by presenting workable solutions to mitigate the problem. Further, a summary of each chapter is described below.

Chapter 1 provides a brief introduction to energy usage and the policies initiated by the Indian government in the building sector. The scope of ECBC 2007 and 2017 and its scope are mentioned in this chapter. The study objectives derived and mentioned are mentioned here. The chapter also discusses the Scope, objectives, and relevance of this study.

Chapter 2 covers the literature survey covering a comprehensive study on the application of renewable energy in the building sector. The scope of the literature survey includes a wide range of subjects dealing with the application of solar PV in the building sector. The use of PV systems and their economics have also been discussed and compared with the conventional grid. The literature survey also discusses the barriers to rooftop solar PV in the building sector.

Chapter 3 discusses the research design and methodology. Based on deductions drawn from various topics in the literature survey it was found that the building energy stars were recently introduced for residential buildings and still not adopted by the existing residential building owners. Based on inferences drawn from the literature survey the research gaps have been identified. The problem statement was formulated.

Chapter 4 explains the concept behind building energy star grading in India. The chapter explores the economic viability of rooftop solar PV for residential buildings in the Indian context through RETscreen simulation software. The chapter also took the subsidy under consideration for analyzing its effect in achieving grid parity. Based on the outcome, Indian states are mapped for grid parity with subsidy and without subsidy scenarios.

Chapter 5 explains the methodology for achieving star grading of existing residential buildings by employing grid-integrated rooftop solar PV. For gauging the energy consumption of an existing building, a case study is undertaken where two years of energy consumption data of residential buildings in Dwarka, New Delhi are analyzed. The chapter also explores with the help of simulations, the technical feasibility and economic viability of rooftop solar PV for residential buildings in different states of India.

Chapter 6 evolved a mitigation methodology process flow. The study concludes that the first step towards mitigating the impact of Electric vehicles (EV) on building energy consumption and its effect on energy star grading of buildings. Different measures were analyzed through simulation software and techno-economic analysis carried out for the cost optimization for upgrading the energy stars for the building in EVs and without EVs scenarios.

Chapter 7 summarizes and concludes the findings. The chapter also provides recommendations on the future scope of work.

CHAPTER 2

LITERATURE SURVEY & CASE STUDY

2.0 PROLOGUE

This chapter encompasses the literature review undertaken for the study. The literature review has been divided into different sections as per the scope of the study. The literature review includes papers from leading journals, government websites, national and international reports on the subject. For the journal paper selections were searched and selected from Google Scholar with the combination of words relevant to the study. In this exercise, more than 150 papers were examined and the same are analysed to find the research gap. The relevant papers which are used for carrying out the study are suitably cited in the relevant chapters.

2.1 SCOPE OF LITERATURE SURVEY

The literature survey for exploring energy-saving measures in the building sector has been undertaken to identify and understand the economics of implementing renewable energy in the building sector. The scope of the literature survey included a wide range of subjects dealing with the application of renewable energy in residential buildings. The following areas were explored in the literature survey: -

- ✓ Global initiatives for establishing building energy codes.
- ✓ India's initiative for establishing building energy codes.

- ✓ Building integrated rooftop Solar PV.
- ✓ Modeling and simulation of Building energy code compliance.

2.2 ESTABLISHING BUILDING ENERGY CODES - A GLOBAL PERSPECTIVE

High energy consumption of existing buildings is noted around the world Therefore, retrofitment of the existing buildings is to be undertaken for curtailing the energy consumption of buildings [6]. A mix of technologies helps in reducing the energy consumption of buildings [7]. Green buildings reduce energy consumption to a great extent but still, the green building market is quite low [8].

Energy Information Administration (EIA) reported that the implementation of energy codes and the latest energy efficiency standards in the US would decrease the energy consumption of the building by 3.6 quadrillions of BTU [9]. A study of the implementation of the building code in Hong Kong suggested that the code not only reduces building energy consumption but also curtails air Pollution [10]. Through Chinese national building standards, public buildings can save 62% of energy [11]. Florida's residential energy code has resulted in a decrease in electricity consumption and a 6 percent decrease in natural gas consumption [12] By implementing of passive energy strategy high-rise apartments in Hong Kong can have an energy savings of 31.4% [13]. As per the study, an energy-efficient building envelope design can result in saving 35% and 47% of total and peak cooling demands respectively [14]. A study in Greece shows that providing insulation for walls, roofs, and floors reduced energy consumption between 20% and 40% [15]. Energy savings potential measures have many financial; institutional and administrative barriers [16].

Building energy codes are also referred to as "energy standards for buildings", "thermal building regulations", "energy conservation building codes" or "energy efficiency building codes". These codes have been developed to encourage awareness and innovation in building design. In addition, these codes provide a framework for the regulatory body to exercise a degree of control over building design. This policy instrument is adopted by many states for improving energy efficiency in the building sector. The status of the availability of building codes for new buildings across the globe is shown below [17]



Figure 2.1 Availability of building code for new buildings across the globe [17] Building codes and efficiency standards provide the energy efficiency yardstick for retrofitting and other energy efficiency measures for an existing building. The status of the availability of building codes for existing buildings (retrofitment) across the globe is shown below



Figure 2.2 Availability of building code for existing buildings across the globe [17]

A study of the Energy Star building in Arizona found that an energy-certified building consumes 8% less energy [18]. A case study highlighted the effectiveness of the Energy Star-R program effectiveness, implementation of the Florida Building Code (FBC) 1997 and FBC 2001 could save more than 10% and 20% percent energy respectively [19]. A study suggests that the top 25% of the buildings in Singapore are also eligible for Energy Stars recommended by the US [20]. Office buildings with Energy Star demand nearly 3 to 5% higher rental premium and with double certification, this premium further increases to 9% [21]. In Egypt, rooftop PV can reduce energy bills in the range of 19% to 61% [22]. A simulation study for a residential building in a Central American tropical location establishes that a passive cooling design strategy significantly reduces the risk of overheating the building [23]. A simulation study reported cooling demand savings of between 26% and 33% by employing passive cooling strategies [24].

2.3 INDIA'S INITIATIVE FOR ESTABLISHING BUILDING

The electricity generation and distribution framework for India is provided by its Electricity Act 1910. After independence, the country's new policies were made to provide impetus to social progress and development by ensuring a supply of electricity, oil, and coal. Energy conservation started in 1970 with a focus on reducing the import of petroleum by reducing consumption. The Working Group outlined the National Energy Policy, and the report was submitted by the group in 1979 which presented the energy scenarios encompassing future energy requirements, in addition, the report also suggested measures to optimize energy usage in India. In 1981, the Inter-Ministerial Working Group was formed for the Energy Conservation drive in the country. The group carried out energy audits of different sectors and estimated that Rs19.25 billion could be saved by investing in energy-saving technologies. In 2001, the Energy Conservation Bill was enacted by the Indian parliament, and the Bureau of Energy Efficiency (BEE) was constituted in 2002 with the mandate of steering energy efficiency in the country through different policies. The important milestones between 1970 and 2020 in the energy conservation initiatives for India are illustrated in Figure 2.3.



Figure 2.3 India's initiative in energy conservation [25]

Between 2005 and 2015 the residential and commercial building floor area doubled and it is estimated by 2030, it will increase to 7.0 billion m² and 1.5 billion m^2 respectively [8]. To enhance energy efficiency in the commercial building sector, the Government of India (GoI) promulgated the Energy Conservation Building Code (ECBC) in 2007. By implementing ECBC suggestions, small buildings can save up to 40% on energy consumption. The energy-saving potential from thermal adaptation is around 16% [26]. Implementing ECBC in Gujarat buildings could save 419,800 GWh between 2010 and 2050, which is equivalent to adding 134,400 MW of electricity generation. Parallel implementation of ECBC in residential buildings in addition to commercial buildings can result in savings of 193,700 GWh and \$32 billion [27]. A study estimated that merely changing the set point of airconditioning and changing the lighting design in all new commercial buildings could result in savings of 7500 GWh/year. Window design, shading, and providing wall and roof insulation as per ECBC norm could yield additional savings of 5900 GWh/year and 450 GWh/year respectively [28]. In 2012, the energy efficiency labeling requirement for a five-star energy air-conditioner was 3.1 (EER). This is 2017 and corresponds to a one-star air-conditioner. The EER is dynamic and revised after a period as a result the average air-conditioner EER has improved from 2.5 in 2010 to 3.3 in 2017.

In the year 2018, sixty percent of room conditioners and a hundred percent of ceiling fans sold in the market were 5-star gradings. These star-labeling appliances have resulted in a saving of 121.29 billion units since 2011[29]. India's Draft National Electricity Policy (NEP) entailed the provision of electricity at an affordable cost to all its citizens. The government intends to reduce the emissions intensity by 33% - 35% by the year 2030 taking the year 2005 as the benchmark and it is also aiming to have 40% of its electricity production from renewable energy resources thus reducing the fossil fuel-based dependency on electricity [30].

ECBC is voluntary where state and Urban Local Bodies (ULB) must play a larger role in implementation in their area of jurisdiction. As per the data available till 2017, the code was promulgated by nine states and it had been adopted by ULBs of two states only [31]. Due to no incentive attached to the building code and customer misalignment of total ownership costs most of the ULBs have avoided the enforcement of ECBC as part of regulatory requirements [32] If the ownership of the ECBC implementation is taken by the ULBs, it will improve property tax and will also open and bring new businesses to cities. Research with the cost-benefit analysis of the implementation of ECBC would educate residential building customers and this will give impetus across 4000 ULBs in India for the adoption of ECBC [33]. A study of different categories of commercial buildings in Jaipur

City (India) predicted that the implementation of ECBC can save energy by up to 42% [34].

In Ahmadabad City, a building can reduce its cooling load by 31% through the implementation of the ECBC guidelines for envelope design [35]. A similar study of ECBC implementation in the hotel buildings in Jaipur shows that buildings can easily save energy in the range of 18.42-37.2%, whereas with the implementation of advanced energy-saving technology, a building can save energy in the range between 53.92-61.75%. The invested amount can be recovered within 2.39 to 6.41 years, whereas for advanced energy-saving technology measures it is between 4.22 and 5.11 years [36]. The simulation result predicted that by the implementation of energy conservation measures recommended by ECBC, small buildings in India can save up to 40% energy as compared to the buildings built as per the business-as-usual practice [27]. Passive measures for the compliance of the build energy codes depend on are most cost-effective as they require minimal or zero investment.

Planting trees in the area near to building reduces the heat gain of the building envelope as nearly 10% of solar radiation is reflected by leaves and absorb about 70%. the building envelope is heated by the remaining 20% of solar energy [37].

In a study, it was found that on weekends an average of 23% of the buildings' energy is wasted in the unoccupied part of the building [38]. Adaptive temperature is a function of environmental temperature and a substantial amount of energy can be saved by fixing the thermostat temperature to adaptive temperature. For the tropical and humid environment of Thailand, the adaptive temperature range for naturally ventilated and air-conditioned office buildings should be 27.4°C and 24.7°C respectively [39]. A study of office buildings in Japan suggested that the thermal comfort temperature should be 27.2°C [40].

In a similar study for the air-conditioned office buildings in Miami and San Francisco city of USA, the thermal set-point was found to be 24°C and 21°C for summer and winter respectively [41]. In another lab-based study of air-conditioned offices for seven different climatic zones of the USA, the thermal set-point was found to be 24°C and 21°C for summer and winter respectively [42]. A lab study of residential buildings in Bangalore, India inferred different thermal comfort temperature is between 20° C – 23.33° C and 22.22° C – 26.66° C for winter and for summer respectively [43]. In an experiment for residential buildings in Chennai thermal comfort band is between 27.6° C – 30.5° C and 26° C – 31.8° C for winter and summer respectively. [44]. For Hyderabad, the adaptive temperature range is between 26° C – 32.45° C; whereas, for the same city ASHRAE suggested 23° C – 26° C and 21° C – 23° C for summer for winter respectively [45].

Deductions: A literature review on Indian building code ECBC 2017 has established that code compliance results in substantial energy saving, and code compliance is economically viable. However, no study is available for converting existing government buildings to ECBC-compliant through retro-fitment. The Indian initiative for implementing the codes is at its nascent stage. Considering the enormous potential available in this area, concerted efforts must be made to exploit this vast untapped potential through optimization of the methods available.

2.4 BUILDING INTEGRATED ROOFTOP SOLAR PV

A study predicts that the EU rooftops could be exploited for the generation of energy. Installation of rooftop solar PV could generate 680 TWh of electricity annually which is nearly one-fourth of electricity consumption and it is costeffective for two-thirds of the consumers [46]. A similar study in Andalusia (Spain), suggests that rooftop solar PV could meet three-fourths of residential energy [47]. Lethbridge, a Canadian city has a rooftop solar PV that has the potential of 300 GWh/year which can meet 38% of its annual electricity consumed and is a better economic option as compared to the conventional grid [48]. In the US, rooftop solar PV potential was analyzed for 51 cities, and with state subsidy 18 could match the grid supply [49]. A similar study for the city of Al-Khobar in Saudi Arabia recommends that by merely utilizing onefourth of the solar PV roof area, the residential apartment can cater for onefifth of its electricity, and shading by PV panels reduces cooling load by 2% [50]. Socket parity by solar PV is not been achieved by the majority of the US states and only six states with subsidies could achieve socket parity [51]. A study establishes that a grid-connected residential solar PV system is feasible in Malaysia [52] A study establishes that a 5 kWp solar PV on a residential building system in Egypt can meet nearly two-thirds of the energy requirement [53]. A study of the rooftop solar photovoltaic potential for Mumbai (India) suggests that it can meet 12.8–20% of the daily energy demand [54]

The ECBC implementation in the Indian different locations countries into composite, hot & and dry, warm and humid, cold and temperate climatic regions, and the availability of global solar radiation for India is given below.



Figure 2.4 Global solar radiation map for Indian states [55]

A Study establishes that 6.4 kW rooftop solar PV in Ujjain city (India) in addition to meeting consumer demand can also feed surplus energy of 8450 kWh/year into the grid [56]. Roof-based cooling load is reduced between 73% and 90% by commissioning a rooftop solar PV system [57]. The invested amount on the installation of a 110 kWp rooftop solar PV system in the city of Bhopal (India) can be recovered within 8.2 years [58]. The net Present value of a grid-connected solar PV system becomes zero for the installation rating between 1.8 kWp and 3.4 kWp, the cost of energy is inversely proportional to the rating of the rooftop solar PV system [59].

Deduction: In India, rooftop solar PV has great potential in bridging the energy gap between desired star-building energy consumption and present energy consumption. However, its economics in different states remain unexplored.

2.5 MODELLING FOR BUILDING ENERGY CODE COMPLIANCE BY SIMULATION

Modeling and simulation help in understanding the building energy performance and provide flexibility for achieving norms of regulatory documents such as ECBC in India's context. Building simulation models are simplifications of reality however the same needs to be validated [60]. Different scenario modeling through simulation helps in the energy analysis of a building which helps managers in communicating reliable information to stakeholders at the operational and strategic levels of organizations [61]. During design-phase energy models usually underestimate the energy consumption by 36% however by incorporating calibration steps, this error can be minimized to 7%. Model calibration helps in improving simulation predictions for the instrumented energy consumption of the building. Calibration steps include plugging updated weather files and revising unregulated and regulated plug-loads of buildings [62]. Occupant behavior varies in a wide range and modeling indicates that most assumptions about occupants need careful examination. In most cases, participants are not aware of the reason for their role in occupant modeling but participants are ready to undergo training [63]. Building Energy Modeling can be made effective if it is incorporated from the design stage itself, as it provides the most desirable and

cost-effective solution for energy-efficient design, and its integration with the building design process is also ensured for the optimized results [64]. The energy performance of Australian homes as compared to international home samples built before 2006 was inferior. US DTS buildings with strict code requirements for building insulation for roofs and windows result in higher-energy performance designs [65].

Urban Building Energy Modeling (UBEM) is a tool used by architects for exploring urban design scenarios and is useful in incorporating building energy during project planning [66]. A comparison of different lighting models including EN 15193:2007 is carried out and found energy savings are of big differences between simulation and EN 15193:2007 [67]. The energy-Plus model's energy consumption estimates are not accurate. Few other models estimate building annual energy consumption which is within 1% of actual measured building energy data; however, modeled homes failed to meet ASHRAE criteria of the calibrated model when compared with hourly energy data [68].

In a study, new homes in eight US states were analyzed for energy code compliance and energy savings potential. The study demonstrates that the total energy performance of new homes for a large population can be gauged by the framework proposed [69]. Impacts of code-compliance on operation costs for housing in Edmonton and Alberta carried out using HOT 2000 simulation. The study indicates that by carrying out upgrades buildings can reduce energy its energy consumption by 12% [70]. Simulation by the Global Change
Assessment Model (GCAM) estimated that implementation of ECBC in the state of Gujarat, India would reduce building electricity consumption by 20%. The reference building is developed for India and it is established that the main energy conservation measures for buildings that are occupied for 8 hours would be through building envelope whereas for buildings that are occupied for 24 hours, building energy conservation from internal loads which include occupancy, lighting, and plugged electrical appliances. The energy performance index (EPI) of reference office buildings is better than buildings abiding by ECBC 2017 norms [71]. Electrical System Estimation and Costing Tool (ESECT) which is an add-in tool of Autodesk Revit can estimate the cost of electrical system construction for residential buildings and monthly energy bills. This can be further analyzed for electrical system modification and cost reduction [72]. Daylighting for a residential apartment in the city of Mumbai was analyzed and it has been found that a building with a southeast orientation and 20% window-to-wall ratio could save up to 26% of the energy required for lighting [73]. A study of a 30 kWp rooftop PV system shows that actual energy generated and simulated energy generation results correlate by 0.99 [74]. In a similar study for 1 MWp solar PV with simulation software PVGIS, PV Watts, and PV Syst simulation tools, errors with the actual are in the range of 5%-30% [75].

Deduction: Modeling and simulation tools bring out the energy leakages and scope of energy conservation for the building under study. Modeling and simulation can also be used for ascertaining building and energy code compliance.

2.6 EPILOGUE

The literature review suggests that the techno-economic study of the implementation of building energy codes for residential buildings is not carried out in the Indian context even though it consumes a large amount of energy. The impact of charging of EV at home is not encapsulated in the residential building energy star labeling program promulgated by the Bureau of Energy Efficiency (BEE), a regulatory body of India with the mandate of improving energy efficiency in the country. Similarly, Rooftop solar PV has been pushed by the government by providing an upfront 30% subsidy on installation on residential buildings, however, the status of grid parity of rooftop solar PV for different states of India is not available.

CHAPTER 3

RESEARCH DESIGN

3.0 **PROLOGUE**

Based on the literature review of the previous chapter there is a research gap for the residential building star energy program in the Indian context. The study is focused on assessing the techno-economics of the Energy Star label promulgated by the Bureau of Energy Efficiency (BEE) for residential buildings. This chapter encompasses the problem statement which is further broken down into objectives and sub-objectives. The motivation for the study is also briefly captured in one of the sections. The research methodology adopted to address the problem statement is incorporated in this chapter.

3.1 RESEARCH GAPS

Based on deductions drawn from various topics in the literature survey it was found that there are immense opportunities for implementing energy codes for residential buildings. Star labeling program for residential buildings is introduced in India to improve energy efficiency in the residential buildings sector. Based on deductions from the literature survey the following research gaps have been identified. \checkmark Grid parity of rooftop solar PV has not been analyzed in the Indian context. The impact of subsidy also needs to be examined in attaining grid parity in different states of India.

- Existing residential buildings in Indian cities have not been investigated for energy consumption and have not been benchmarked against the proposed Energy Star label. The techno-economic potential of rooftop solar PV for attaining high-energy stars has also not been explored by any study.
- ✓ The Energy Star label to the residential building is awarded against the notified standard by the regulatory body and Electric Vehicle (EVs) has not been catered to as a plugged load for the residential buildings. Its impact on existing energy star labeling of Indian residential buildings needs to be quantified. The research needs to be further extended so that increased load due to EVs is mitigated by appropriate energysaving measures.

PROBLEM STATEMENT: It is imperative to understand the "rooftop solar PV grid parity" for undertaking the feasibility studies of deploying rooftop solar PV in attaining high energy stars by existing residential buildings.

3.2 MOTIVATING FACTOR

From the literature survey, it emerged that there is a scope for large energy savings in the residential buildings sector. The Indian government has recently introduced Energy Star labels for residential building energy. Existing buildings can undergo retrofitting and employ renewable energy to achieve the desired energy stars. Techno-economic feasibility of solar PV reported for different countries including India by different studies but the same is not available for rooftop solar PV in Indian cities. The problem statement has not been addressed; building energy star labels and grid parity can complement each other.

3.3 OBJECTIVES

The following are the objectives to address the problem statement: -

Modeling of an existing government building for ECBC 2018-R compliance.

Sub Objectives incidental to the main objective are: -

(i) Simulation of the different scenarios for attaining residentialEnergy star grading and ascertaining the viability of the retrofitment.

(ii) Techno-economic assessment of renewable energy for residential buildings technologies for achieving Five-star energy grading

3.4 METHODOLOGY AND DATA COLLECTION

The study undertakes a two-pronged approach to address the live problem of achieving the five-star energy label of an existing residential building in Indian cities. In the first approach deployment of rooftop solar PV is explored. In the second approach, additional passive energy-saving measures are to be explored. A detailed research methodology followed is given below.



Figure 3.1 Research Methodology for rooftop solar PV

Energy Consumption data taken from a residential society in Palam, New Delhi was taken for benchmarking the energy consumption of residential consumers. RETscreen an open-source software is used for the rooftop solar PV for the residential building under study. The energy-saving potential by building envelope retrofitting and change in the appliance is undertaken by the eQUEST software. The impact of EV and on residential buildings and its impact on Energy Star is also incorporated in the study. The economics of the energy mitigation method considered is also analyzed using eQUEST. The methodology adopted for this is given in Figure 3.2



Figure 3.2 Flow chart for the study of EV scenario

3.5 EPILOGUE

A quantitative approach is undertaken to gauge the existing residential buildings' energy consumption and find out the energy gap for the Energy Star label. RETscreen and eQUEST software are utilized for the simulations of different scenarios. The study aims to evaluate the grid parity of rooftop solar PV in different states of India its scope bridging the energy gap of the existing residential buildings so that the desired energy star label can be achieved. The impact of EVs on the energy star of a building and its mitigation by rooftop solar PV and building envelope retrofit need to be evaluated.

CHAPTER 4

STATUS OF ROOFTOP SOLAR PV GRID PARITY FOR RESIDENTIAL BUILDINGS IN INDIA

4.0 PROLOGUE

This chapter explores the status of rooftop solar PV grid parity for residential buildings for different scenarios. The government of India extends upfront a 30 % subsidy on the installation of rooftop solar PV up to 3 kWp. The residential building energy tariff is a state-subject and therefore the viability varies from state to state. This chapter takes the stock of grid parity of rooftop solar PV across all the states of India. RRTscreen simulation for 3 kWp and 2 kWp with and without subsidy and for major cities across is carried out to gauge the grid parity.

4.1 GRID PARITY CONCEPT

To overcome the increasing demand for energy and more stress on clean energy, the penetration of solar PV is increasing in the global market. The cost of conventional coal energy resources is increasing whereas PV cost is reducing. India's government aims to install 175 GW of renewable energy by 2022. In this target, solar energy will lead and contribute 100 GW, and 40 percent of this would be from rooftop solar PV. India's solar PV installation has already crossed 28 GW as of March 2019 [36]. The status of the installation of grid-connected solar PV in India is shown in Figure 4.1. Rooftop solar PV installation in India has touched 5.4 GW in December 2019. The distribution of rooftop solar PV in the different sectors is shown in Figure 4.2.



Figure 4.1 Status of grid-connected solar PV in India [76]



Figure 4.2 Installation status of rooftop solar PV in the different sectors [77]

Reduction in the cost of PV modules in the Global markets has made solar PV economically equivalent to fossil-based grid supply, which is termed grid/ socket parity. Grid/socket parity is a condition where the cost of energy generation from solar PV is cheaper than conventional grid electricity. In

economic terms, if the levelized cost of electricity from solar (energy source) is less than the grid cost then the energy source achieved static grid parity and in dynamic grid parity, the NPV of the solar PV system is compared to the NPV returns generated by the solar PV system.

4.2 COST OF ENERGY FOR RESIDENTIAL CONSUMERS IN INDIAN STATES

Energy tariff for residential consumers is telescopic where the cost of energy increases as per the increase in energy consumption. Accordingly, the residential energy consumers can be grouped into low, moderate, and high as per energy consumption, and the same is tabulated in Table 4.1. Electricity tariff for medium and high residential consumers for selected Indian states is tabulated in Table 4.3. The electricity tariff escalation rate is evaluated by the Compound Annual Growth Rate (CAGR) from 2009-10 to 2018-19 for the medium (400kWh/month) and high (1000kWh/month) energy consumers from the Centre of Electricity Authority (CEA) annual report [18]. The calculated CAGRs of energy for medium and high residential urban consumers are 5% and 5.4% respectively.

| Category | Classification | of | Monthly | Electricity |
|----------|----------------|--------|--|-------------|
| | | | 5 | - |
| | | | | |
| | residential | energy | Consumption | |
| | | | - · · · · I | |
| | | | | |
| | consumers | | | |
| | | | | |
| | | | | |
| Α | Low | | E<200 kWh | |
| | | | | |
| | | | | |
| В | Moderate | | 200 <e <400="" kwh<="" td=""><td></td></e> | |
| | | | | |
| | | | | |
| С | High | | 400KWh <e< td=""><td></td></e<> | |
| C | 111911 | | | |
| | | | | |

Table 4.1 Residential energy consumer groups

| State | Energy tariff | Electricity tariff |
|----------------|----------------------|--------------------|
| | Rs /kWh | Rs /kWh |
| | (Moderate consumers) | (High consumers) |
| Andhra Pradesh | 7.50 | 8.50 |
| Bihar | 6.67 | 6.67 |
| Delhi | 4.50 | 6.50 |
| Gujarat | 5.20 | 5.20 |
| Haryana | 6.30 | 7.10 |
| Karnataka | 7.80 | 7.80 |
| Maharashtra | 10.36 | 11.82 |
| Madhya Pradesh | 6.50 | 6.50 |
| Orisa | 5.30 | 5.70 |
| Punjab | 7.30 | 7.30 |
| Rajasthan | 7.65 | 7.95 |
| Tamil Nadu | 4.60 | 6.60 |
| Telangana | 8.50 | 9.0 |
| Uttar Pradesh | 6.50 | 7.0 |
| West Bengal | 6.64 | 7.0 |

Table 4.2 Energy charges for residential consumers [78]

4.3 EVALUATION OF ENERGY GENERATION FROM ROOFTOP SOLAR PV IN INDIAN CITIES

MNRE extends a subsidy on the installation of rooftop solar PV which is 40% for solar PV plants of rating 3kWp and below however this subsidy is reduced to 20% for solar PV plants rating between 3kWp and 10kWp [20]. A residential rooftop PV system of 2 kWp and 3 kWp is considered for moderate and high energy consumers respectively. In the Indian rooftop solar PV market, Tata Solar has been the leader for the last seven years (2014-21). Tata solar rooftop system (3 kWp and 2 kWp) is considered for the study and the cost of the same is tabulated in Table 4.3.

| Rating | Cost in the absence of subsidy | Cost including subsidy (Rs) |
|--------|--------------------------------|-----------------------------|
| | (Rs) | |
| 3 kWp | 1,90,000 | 1,40,000 |
| 2 kWp | 1,40,000 | 1,05,600 |

Table 2.3 TATA solar rooftop solar PV system cost [79]

Energy generation cost from the rooftop solar PV system is calculated for the 34 Indian cities by using RETScreen version 10, a software program developed by Natural Resource Canada. The energy simulation included both the scenario (subsidy and without subsidy) for 3 kWp and 2 kWp rooftop solar PV. The results obtained for the 3 kWp and 2kWp rooftop solar PV and its comparison with grid energy charges are shown in Figure 4.3 to Figure 4.6



Figure 4.3 Grid parity status of 3 kWp rooftops solar PV in Indian cities (including subsidy)



Figure 4.4 Grid parity status of 3 kWp rooftops solar PV in Indian cities (in the absence of subsidy)



Figure 4.5 Grid parity status of 2kWp rooftop solar PV in Indian cities (including subsidy)



Figure 4.6 Grid parity status of 2kWp rooftop solar PV in Indian cities (in the absence of subsidy)

Result analysis of 3 kWp rooftops solar PV (with subsidy) indicates that grid parity is achievable in most of the cities, in 24 cities out of 34 cities energy generation cost of 3 kWp rooftops is less than the grid. However, if the state subsidy is removed the economic viability of 3 kWp reduces drastically and only 12 cities could have grid parity. Among the 34 cities, Jaipur city in Rajasthan state has the minimum energy cost of Rs 5.60/kWh, whereas in other cities this varies in the range of Rs 6.50/kWh- Rs 7.00/kWh. Because of the high energy tariff for residential consumers and the availability of high solar radiation throughout the year, Maharashtra state leads other Indian states in the grid parity. In addition to Maharashtra, Rajasthan, Telangana, Karnataka, and Andhra Pradesh could also breach achieve grid parity barrier even in the absence of the subsidy.

Result analysis of 2 kWp rooftop solar PV (including subsidy) simulation result indicates that with the reduction of the rating of rooftop solar PV grid parity reduces in Indian states as only 5 cities out of 34 cities could achieve grid parity. In the absence of subsidy, no Indian state could achieve grid parity as electricity tariffs for moderate energy residential consumers are low as compared to high energy consumers. Maharashtra and Rajasthan are the only two states which could only achieve grid parity for a 2 kWp rooftop with the state's 40% subsidy. Jaipur city recorded the lowest energy cost for 2 kWp rooftops solar PV i.e. Rs 6.68/kWh, whereas in most other Indian cities energy cost varies in the range of Rs 7.50/kWh-Rs 8.50/kWh.

Economics analysis of 2kWp and 3 kWp rooftops solar PV carried out by evaluation of payback period, NPV, and IRR. The payback period is calculated by dividing the invested amount on the installation of rooftop solar PV by the annual saving in energy charges. For calculation of NPV and IRR in addition to the cost of rooftop solar PV, Operation and Maintenance (O&M) cost and recurring cost for replacement of the inverter in the 13th year was also plugged in RETsceeen software. Other economic parameters such as escalation in utility electricity prices considered for NPV calculation are tabulated in Table 4.4.

| Parameter | Plugged | References |
|------------------|----------|------------|
| | value | |
| Discount rate | 9.36% | [80] |
| life of solar PV | 25 years | [80] |
| Debt/Equity | 70/30 | [80] |
| ratio | | |
| Interest on Debt | 10% | [81] |
| Debt period | 10 yrs | [80] |

Table 4.4 Parameters plugged in the RETsceen module.



Figure 4.7 Payback period of 3 kWp rooftops solar PV in Indian cities



Figure 4.8 Payback period (Years) of 2 kWp rooftops solar PV



Figure 4.9 IRR Figure (Percent) of 3 kWP rooftop solar PV



Figure 4.10 IRR (Percent) of 2kWp rooftop solar PV

3 kWp rooftop RET simulation results show with the help of state subsidy 25 cities out of 34 cities IRR is more than 11%, the rate at which solar rooftop PV owners can easily obtain a debt from any financial institution. In the absence of subsidy, 8 cities could match this competitive IRR. The economic performance of rooftop solar PV is best in Pune city in which the highest IRR and NPV (with subsidy) of 26.4% and Rs 4,32,121 is achieved. The economic performance of rooftop solar PV is worst in Ahmadabad city where IRR and NPV (with subsidy) of 0.99% and Rs -12,174 is achieved. In the absence of the subsidy, the economic performance of 3 kWp rooftops solar PV is degraded, and IRR and NPV for Pune and Ahmadabad are reduced to 18.7%, Rs 3, 79,589, and -2.1%, Rs -64,178 respectively.

2 kWp rooftop RET simulation results show with the help of state subsidy, 8 cities out of 34 cities IRR is more than 11%. In the absence of subsidy, no city could match this competitive IRR. Economic performance rooftop solar PV is best in Pune city in which the highest IRR and NPV (with subsidy) of 14.4% and Rs 1,45,162 is achieved. The economic performance of rooftop solar PV is worst in Ahmadabad city where IRR and NPV (with subsidy) of -9.19% and Rs -59,235 is achieved. In the absence of the subsidy, the economic performance of 3 kWp rooftops solar PV is degraded, IRR and NPV for Pune and Ahmadabad reduced to 10%, Rs 1,09,345 and -11.3%, Rs -95,385 respectively. As per the RETscreen simulation results discussed above the cities can be graded as most favorable, preferred, and not favorable for 3kWp and 2 kWp rooftops solar PV in terms of grid parity. Economic parameters employed for this classification are tabulated in Table 4.5. To obtain the most favorable tag in terms of grid parity the city must satisfy all the parameters (energy generation/payback period/NPV). Based on this grid parity map of the country for 3 kWp and 2 kWp rooftops solar PV with subsidy/ without subsidy is shown in Figures 4.11,4.12 and 4.13.

| Favorability | Energy Cost | Payback | NPV (Rs) |
|--------------|----------------|-------------|-----------------|
| Index | (Rs /kWh) | Period | |
| | | (years) | |
| Green (Most | Less than grid | Less Than 6 | More than 50% |
| Favorable) | tariff | | capital cost |
| Yellow | Within 10% | 6-7.5 | Between 50%- |
| (Favorable) | of the grid | | 25% of capital |
| | tariff | | cost |
| Red (Not | More than | More than | Less than 25% |
| favorable) | 10% of the | 7.5 | of capital cost |
| | grid tariff | | |

Table 4.5 Parameters for Grid parity for rooftop solar PV for Indian states



Created with mapchart.net

Figure 4.11 3 kWp Rooftop solar PV grid parity with subsidy



Figure 4.12 3 kWp Rooftop solar PV grid parity without subsidy



Created with mapchart.net

Figure 4.13 2 kWp Rooftop solar PV grid parity with subsidy

4.5 EPILOGUE

Simulation results in this chapter show that widespread grid parity across the states of India for 3 kWp and 2 kWp rooftop subsidy is achieved with the state subsidy. However, in the absence of subsidy, the grid parity of rooftop solar PV is reduced. Grid parity is largely the function of existing grid tariff in the state and solar radiation available in the state. The states are also subsidizing the grid supply to the residential consumer where consumption of consumer is less the 100-200 kWh/month. Therefore, the cities with similar solar radiation have different positions in grid parity. Rooftop solar PV is a viable energy solution for residential buildings and can be explored for higher energy star grading of buildings

CHAPTER 5

STUDY OF RESIDENTIAL BUILDINGS FOR ENERGY STARS LABELING

5.0 PROLOGUE

Energy building codes are in vogue across the world and in alignment with the energy efficiency drive India has initially promulgated energy building codes for commercial buildings. As an extension of this building code, a residential building code is also available which is voluntary for the consumers. This chapter analyses the energy consumption by residential buildings through a case study of residential buildings in a society in New Delhi and benchmarks against the Energy Star label. RET screen simulation carried out for seven major cities of the country for assessing the techno-economics potential of rooftop solar PV is explored to bridge the energy gap for the desired Energy Star label.

5.1 RESIDENTIAL BUILDINGS STAR LABEL

After the success of the star labeling program for electrical appliances, the Bureau of Energy Efficiency (BEE) introduced the star labeling program for all single and multiple-dwelling residential buildings in 2019. For a residential consumer to avail of this label there is no minimum requirement for the building area or plugged load (kW). The Energy Performance Index (EPI) of a building (annual energy consumption in kilowatt-hours per square meter of the building) is the parameter considered for awarding the star label of the building. The EPI of the building consists of three E1, E2, and E3 components. E1 and E2 include building envelope, Lighting system, and Air conditioning system (AC). For the building only 25% of the space is assumed to be air-conditioned with 24 deg C. EPI (E3) for other building electrical appliances Such as water pump, washing machine, microwave oven, grinder, refrigerators, TV, etc. is considered in the range 7 to 9. Star labeling for different climate regions is tabulated in Table 5.1 considering the value of E2 as 8.

| Energy Stars | Composite | Warm & Humid | Hot and Dry | Temperate |
|-----------------|------------|-----------------|-----------------|-----------------|
| * | 60 < X ≤68 | $66 < X \le 72$ | 63 < X ≤75 | $36 < X \le 39$ |
| ** | 53 < X ≤60 | $57 < X \le 65$ | $55 < X \le 63$ | 32 < X ≤36 |
| *** | 45 < X ≤53 | $47 < X \le 57$ | $46 < X \le 55$ | $29 < X \le 32$ |
| * * * * | 37 < X ≤45 | $38 < X \le 47$ | $37 < X \le 46$ | $25 < X \le 29$ |
| * * * * * | X ≤ 37 | X ≤ 38 | X ≤ 37 | X ≤ 25 |

 Table 5.1 Star labelling requirement [EPI(X)] in different climate zones [82]

5.2 CASE STUDY FOR QUANTIFYING THE ENERGY CONSUMPTION OF A RESIDENTIAL BUILDING

For ascertaining the energy consumption of residential buildings study has been undertaken for residential buildings in the Palam area of New Delhi, India details of the location are given in Table 5.2. The layout of flats as per the RETscreen location module is given in Figure 5.1.

| 77.1 ⁰ E |
|---------------------|
| 1B -Very Hot-Dry |
| |

Table 5.2 Building Location details



Figure 5.1 Layout of flats as the RETscreen

This government residential campus has 81 building blocks, and each block has four houses. The residential block is a two-story structure consisting of four houses with two basement parking. Each house has a similar building area and carpet area of 314 m^2 and 628 m^2 respectively. This study aims to gauge the techno-economic potential of rooftop grid-connected solar PV on residential buildings and achieve five-star energy labeling for the building.

5.3 BENCHMARKING OF BUILDING ENERGY CONSUMPTION

Electricity consumption data was collated from April 2017 to March 2019 for the residential buildings. Annual energy consumption is used to distribute the houses in groups and the same is shown in Figure 5.2.



Figure 5.2 Energy consumption of residential consumers

An increase in the average annual energy consumption for residential buildings was observed from 2017-18 to 2018-19. In 2017-18 the annual energy consumption which was 7236.72 kWh increased to 8101.34 kWh in 2018-19. The commercial grid supply to the residential buildings under study is from BSES Rajdhani Power Limited (BRPL) and this is the sole electricity distribution company for south and west Delhi. The energy tariff for the year 2019-20 applicable to residential buildings is tabulated in Table 5.3.

| Slabs | Energy const | Imption Energy Tariff (| Rs/kWh) |
|-------|--------------|-------------------------|---------|
| | (kWh) | | |
| A | up to 200 | 3.00 | |
| В | 201–400 | 4.50 | |
| С | 401-800 | 6.50 | |
| D | 801–1200 | 7.00 | |
| E | over 1200 | 8.00 | |

Table 5.3 Energy tariff for Delhi residential buildings[83]

As per the annual energy consumption, the residential energy consumers are classified into four classes that are tabulated in Table 5.4.

Table 5.4 Residential consumer classification based on annual energy consumption.

| Class | Consumer | Annual Energy Consumption (kWh) |
|-------|-----------|---------------------------------|
| А | Low | E≤4800 |
| В | Moderate | $4800 < E \le 9600$ |
| С | High | 9600 <e≤ 14400<="" td=""></e≤> |
| D | Very High | E > 14400 |

By analyzing the annual energy consumption for the years 2017-18 and 2018-19, more than three forth of the residential consumers' energy consumption is moderate and by suitable energy saving measures, the required five-star energy star label for the building can be achieved. Residential consumers categorized into different classes based on annual energy consumption for 2017-18 and 2018-19 are shown in Figure 5.3 and Figure 5.4 respectively.



Figure 5.3 Categories of consumers based on annual electricity consumption (2017-18)

Analysis of the annual energy consumption two years data of the residential consumer it is clear that the energy consumption has increased for all the classes. As a result, there is a 10% reduction in the moderate energy consumer category, and the low-energy consumer category reduced from 14% to 10% whereas, the high-energy consumer category increased from 8% to 23% in the same period. Therefore in this study, high energy consumption is considered as the target energy consumption which needs to be addressed by suitable energy-saving measures so that it can achieve the desired 5-star energy label.



Figure 5.4 Categories of consumers based on annual electricity consumption (2018-19)

5.4 EVALUATION OF ENERGY GENERATION BY ROOFTOP

SOLAR

In this section, energy generation capacity by rooftop solar PV is carried out based on the area available for the installation of rooftop solar PV, and state subsidy extended to residential buildings consumers for the installation of rooftop solar PV. As per the availability of rooftop space, solar of different capacities are considered for the installation. Energy generation of rooftop solar PV is evaluated using a rooftop solar calculator hosted on the website of the Ministry of New and Renewable Energy (MNRE). The results obtained are tabulated in Table 5.5.
| %Roof | Roof area | Rooftop | Electricity | Solar PV Plant |
|-------------|-------------------|--------------------|-------------|----------------|
| utilization | (m ²) | Solar PV | generated | cost (Rs) |
| | | (kW _p) | (kWh/yr) | |
| 20 % | 30 | 3 | 4140 | 75600 |
| 26 % | 40 | 4 | 5520 | 106600 |
| 33 % | 50 | 5 | 6900 | 139400 |

Table 5.5 Energy generation by rooftop- solar calculator [84]

MNRE extends a subsidy on the installation of rooftop solar PV which is 40% for solar PV plants of rating 3kWp and below however this subsidy is reduced to 20% for solar PV plants rated between 3kWp and 10kWp. MNRE benchmark cost is considered for the proposed solar PV plant. The site under consideration receives 5.06 kWh $/m^2/day$ solar radiation and the summary of the availability of horizontal solar as per the RETscreen is given in Figure 5.5.



Figure 5.5 Solar radiation at the study site as per RETscreen

5.5 ECONOMIC EVALUATION OF ROOFTOP SOLAR PV

For assessing the potential of rooftop solar PV MNRE rooftop solar PV calculator is utilised which is available on the Ministry of New and Renewable (MNRE) website. The energy generation obtained from the solar calculator is tabulated in Table 5.6 and the same was validated by using NREL RETScreen simulation software [51]. Energy output from the RET screen is comparable to the MNRE solar PV calculator and within the tolerance of 5 percent.

| (%Roof Utilization) | Area (m ²) | Rating of Rooftop | Annual Energy |
|---------------------|------------------------|-----------------------------|------------------|
| | | Solar PV (kW _p) | Generation (kWh) |
| 20 % | 30 | 3 | 4320 |
| | | | |
| 26 % | 40 | 4 | 5760 |
| | | | |
| 33 % | 50 | 5 | 7200 |
| | | | |

Table 5.6 Rooftop solar PV energy simulation using RETscreen

Ministry of New and Renewable Energy (MNRE) extends a flat 40% subsidy on the installation of rooftop solar PV up to 3 kWp, therefore installation of 3 kWp rooftop solar PV is considered for this study. Installation of grid-tied 3 kWp rooftops solar PV would offer a five-star energy label to eighty percent of the houses. Building energy star distribution post installation of gridconnected rooftop solar PV for 2017-18 and 2018-19 as per the residential energy star label program of India is shown in Figures 5.6 and Figure 5.7 respectively.



Figure 5.6 Energy star building status post installation of 3 kWp rooftops solar PV (2017-18) Residential buildings with low and moderate energy consumption (Table 4.3) could achieve a five-star energy label by the installation of rooftop solar PV whereas other building owners with high/very high energy consumption could achieve an additional two-star energy star.



Figure 5.7 Energy star building status post installation of 3 kWp rooftops solar PV (2018-19) The financial analysis of rooftop solar PV is carried out using the financial module of RETscreen. The net present value (NPV), Internal rate of return (IRR), and payback period were obtained by plugging the relevant value in the RETscreen. The value plugged in the RETscreen is given in Table 5 Chapter 4 and the results obtained are tabulated in Table 5.7

| Rooftop Solar PV | Payback period | NPV (Rs) |
|------------------|-------------------|----------|
| 3 kWp | 6.5 years | 1,40,000 |

Table 3 NPV and Payback period of 3kW rooftop solar PV

This exercise is further extended to ascertaining the feasibility of rooftop solar PV in the other cities of India located in different climatic zones, the RETscreen simulation carried out in respect of other cities and energy generation obtained by rooftop solar PV are tabulated in Table 5.8, economic results of rooftop solar PV for these cities obtained from the financial module of RETscreen are given in Table 5.9.

| City | Location | Daily solar | Energy tariff |
|-----------|----------------------------|--|----------------|
| | [latitude(°)/longitude(°)] | radiation (kWh/m ² /day) | (Rs /kWh) [56] |
| Mumbai | 19.1 / 72.9 | 5.12 | Rs 7.51 |
| Chennai | 12.8 / 80.1 | 5.37 | Rs 6.10 |
| Kolkata | 22.5 / 88.3 | 4.86 | Rs 8.92 |
| Bengaluru | 13 / 77.6 | 5.32 | Rs 7.80 |
| Hyderabad | 17.5/78.5 | 5.00 | Rs 9.00 |
| Ahmedabad | 23.1/72.6 | 5.50 | Rs 5.20 |
| Pune | 18.5/73.8 | 5.52 | Rs 11.54 |

Table 5.8 Energy tariff and solar radiation of Indian cities

| City | Annual Energy | Simple | NPV with 4% annual |
|-----------|---------------|--------------|---------------------------|
| | Generation | Payback | escalation of electricity |
| | (kWh) | Period (yrs) | tariff (Rs) |
| Mumbai | 4274 | 3.5 | 375,956 |
| Chennai | 4330 | 7.3 | 90,872 |
| Kolkata | 4082 | 6.5 | 124,625 |
| Bengaluru | 4334 | 5.2 | 200,711 |
| Hyderabad | 4127 | 4.6 | 242,970 |
| Ahmedabad | 4646 | 9.2 | -12,174 |
| Pune | 4614 | 3 | 432,126 |

Table 5.9Techno-economic results of 3kWp rooftop solar PV

CHAPTER 6

IMPACT OF ELECTRIC VEHICLES ON ENERGY STARS FOR EXISTING RESIDENTIAL BUILDINGS IN INDIA AND ITS MITIGATION

6.0 **PROLOGUE**

Across the globe transport sector consumes 26% of total final energy, and globally this results in 8 Gt of CO2 emissions in 2022. Forty-five percent of fossil fuel demand globally is contributed by road transport and internal combustion cars sales grew by an annual average of 3.4% from 2010 to 2017. is responsible for nearly 45% of global oil demand. To overcome the dependency on fossil fuels countries are migrating to zero-emission vehicles, In 2023, EV (Electric Vehicle) sales were 14 million and every fifth vehicle sold globally was EV. China and Europe's market share of EVs in this period was 60% and 25% respectively. India, the second-largest electric two-wheeler market globally sales grew by 40% in 2023 as compared to 2022.

6.1 EV SCENARIO IN INDIA

Globally EV sales have increased nearly 50 times in the last decade (2012-22) and the 2022 EV sale is 10 percent of global car sales [85]. EV is the sustainable alternative to IC engines in a circular economy [86]. In 2023-24 an additional 90,432 EVs were added, and its growth between 2017-18 and 2023-24 is shown in Figure 6.1 [87]. Electricity

consumption of EVs in city driving conditions is 84 Wh/km and 123 Wh/km for low and high power respectively [88]. Tail-end CO and CO₂ emissions by vehicle emissions reduce by 60 % and 17 % respectively if conventional IC vehicles are migrated to EVs [89],[90]. The best part of EVs is that the charging of EVs at residential buildings matches the valley of energy demand curve [91].



Figure 6.1 Four wheelers EV sales in India (2017-24)

Residential building energy consumption is taken from the study carried out for residential buildings located in Dwarka Delhi, India mentioned in the chapter 5. In the building's energy consumption data, EV is not a plugged load and the same is also not considered by the BEE building energy star label. Government policies have given impetus to the EV, and in the last five years, the sale of EVs has followed an upward trajectory. At present contesting policies of building energy star labeling and EVs are not gelled. The addition of EVs as plugged loads has an adverse impact on the star label of the building. Further sections of the chapter will quantify the impact of EVs on residential energy stars and explore the measures to mitigate this additional EV load. To ascertain the viability of the measures economics is also analyzed in the study.

6.2 EVALUATING EV LOAD ON BUILDING

For assessing the additional electrical load on the building due to EV charging at home TATA Nexon EV is considered for this study. The car is equipped with a 30.2 kWh lithium-ion battery and a fully charged battery provides a range of 312 km under standard conditions. For modelling the EV load on a building a PDF function has been used. The historical data of traveling patterns and vehicle numbers in a commercial building have been used to derive the standard deviation of the PDF function. The graphs have been obtained and Monte Carlo scenarios applied to generate the EV load on a building with charging duration. Probability Distribution Function (PDF) is used to evaluate the EV load profile. Using PDF for the arrival and departure of EVs on weekdays is generated. Equations 1 and 2 show the PDF for the arrival and departure of EVs respectively. The probability and cumulative probability distribution of EV on weekdays are shown in Figure 6.2 and Figure 6.3 respectively.

$$PDF_{Ar} = \begin{cases} \frac{1}{\sigma_{Ar}\sqrt{2\pi}} e^{-\frac{(t+24-\mu_{Ar})^2}{2\sigma_{Ar}^2}}; 0 < t \le \mu_{Ar} - 12\\ else\\ \frac{1}{\sigma_{Ar}\sqrt{2\pi}} e^{-\frac{(t-\mu_{Ar})^2}{2\sigma_{Ar}^2}} \end{cases}$$
(1)

 $d = distance \ travel \ in \ km, \ \sigma Ar$ is the standard deviation of arrival

time and μ_{Ar} is the mean arrival

$$PDF_{Dr} = \begin{cases} \frac{1}{\sigma_{Dr}\sqrt{2\pi}} e^{\frac{(t-\mu_{Dr})^2}{2\sigma_{Dr}^2}}; 0 < t \le \mu_{Dr} + 12\\ else\\ \frac{1}{\sigma_{Dr}\sqrt{2\pi}} e^{\frac{(t-24-\mu_{Dr})^2}{2\sigma_{Dr}^2}} \end{cases}$$
(2)

distance travel in km, σ_{Dr} is the standard deviation of departure time and μ_{Dr} is the mean departure.



Figure 6.2 Probability distribution of arrival and departure of EV



Figure 6.3 Cumulative probability distribution of arrival/departure of EV For the distance covered by EV in a day a PDF function has been created using logarithmic distribution given in equation 3 and the plot obtained for the distribution is shown in Figure 6.4. The State of Charge (SoC) of EV batteries is calculated from equations 4 and 5. The battery charging EV load profile for residential buildings is created from Equation 6.

$$PDF_{Dis} = \frac{1}{d\sigma\sqrt{2\pi}} e^{-\frac{(\ln d - \mu)^2}{2\sigma^2}}; d = distance \ travel \ in \ km \ and \ \sigma \ is \ the \ standard$$

deviation

(3)



Figure 6.4 PDF of daily distance traveled by EV

$$SoC = 1 - \frac{a}{R}$$
; $R = Range \ of \ EV \ on \ full \ charge$ (4)

$$PDF_{SoC} = \frac{1}{R(1-SoC)\sigma\sqrt{2\pi}} e^{-\frac{(\ln(R(1-SoC))-\mu)^2}{2\sigma^2}}$$
(5)

$$T_{charing} = kWh_{battery}(1 - SoC)/(.82 \times P_{charger})$$
(6)

To evaluate the additional EV load, equations 5 and 6 given above are used. The performance of EVs in city traffic is assumed 90% of standard conditions spelled out by the EV manufacturer. The charger installed at residential buildings with an efficiency of 82% will result in an energy requirement of EV 0.132Wh/km. More than 95% of the EV traveled distance in a day is less than 50 Km and considering 100% charge at the residential building will have an additional annual energy demand of 1724 kWh. This additional energy demand will reduce the two energy stars of the building as per the existing energy star label for residential buildings. As per the energy consumption of residential consumers mentioned in Chapter 5, the consumers are categorized into three categories tabulated in Table 6.1. By analyzing the annual energy consumption data of residential consumers mentioned in Chapter 5, it is evident that most consumers' annual energy consumption is below 12,000 kWh. Therefore, this is taken as benchmark consumption, whereas for the building to have a five-star energy label annual energy consumption should be less than 5940 kWh. The energy gap of 7784 kWh/annum exists for the residential building with EV and this gap reduces to 6060 kWh/annum if EV is not available for the five-star energy label.

| | 1 | 1 | 1 | |
|-----------|---|----------------|---------------|----------------|
| Category | Annual | Average energy | Energy Gap | Energy Gap for |
| | Energy | consumption | for Five-Star | Five-Star with |
| | Consumption | (kWh) | without EV | EV |
| | (kWh) | | (kWh/annum) | (kWh/annum) |
| Moderate | 4800 <e td="" ≤<=""><td>7200</td><td>1260</td><td>2984</td></e> | 7200 | 1260 | 2984 |
| | 9600 kWh | | | |
| High | 9600 <e≤< td=""><td>12000</td><td>6060</td><td>7784</td></e≤<> | 12000 | 6060 | 7784 |
| | 14400kWh | | | |
| Very High | E > 14400 | - | More than | More than 7784 |
| | kWh | | 6060 | |

Table 6.1 Consumer categorization based on annual energy consumption.

Residential energy tariff is slabbed in different categories such that a consumer who consumes more would pay more per unit consumed. The telescopic energy tariff is categorized into 5 categories which are tabulated in Table 6.2.

| Monthly energy | Energy Tariff (Rs/kWh) |
|----------------|------------------------|
| consumption | |
| up to 200 kWh | Rs 3.00 |
| 201–400 kWh | Rs 4.50 |
| 401–800 kWh | Rs 6.50 |
| 801–1200 kWh | Rs 7.00 |
| over 1200 kWh | Rs 8.00 |

Table 6.2 Electricity charges for Delhi residential building [80]

6.3 TECHNO-ECONOMIC ASSESSMENT OF ENERGY STAR

APPLIANCES FOR RESIDENTIAL BUILDINGS

The Energy Star appliance retrofit, installation of rooftop solar PV, and building envelope retrofit are the measures that are explored in this section. Economic parameters Payback Period (PP), Net Present Value (NPV), and Cost of Conserved Energy (CCE) are evaluated to ascertain the economic viability of retrofits. These parameters are defined as follows.

PP=Co/ Δ OC where C₀ is the upfront expenditure for procurement of Energy Star label appliance and Δ OC is economic saving due to reduction in the operating cost.

NPV= $\sum_{k=1}^{n} OC/(1 + d)_{k}$ _Co ,Co and OC as defined earlier, 'd' is the discount factor and k is the number of years

 $CCE=Co/\sum_{k=1}^{n} \Delta E/(1+d)_k$ where ΔE is energy savings by the star label appliance over its life (k years) which is discounted every year.

Building energy consumption from the plugged electrical load is primarily from room air-conditioners, refrigerators, lighting sources, and ceiling fans. In addition, water electric heaters, washing machines, televisions, and computers also contribute to the energy meter bill. The replacement of a refrigerator, room air-conditioner, and ceiling fan is considered in this study. Highly energy-efficient appliances with high energy star labels are available with additional premium charges. The operating hours of the appliances and their life are considered according to the climatic regions of India are given below in Table 6.3.

| Appliance | Quantity | Operating | Life of | Energy |
|-------------------------|----------|-----------------|--------------|----------------------|
| | | Hours/year [92] | appliance | consumption/ |
| | | | (years) [92] | EER baseline [78] |
| Room air conditioner | 02 | 1440 | 10 | 2.9 |
| Ceiling Fan | 02 | 1600 | 15 | 128 kWh/year |
| Refrigerator | 01 | 8760 | 15 | 509 kWh/year |

Table 4 Residential electric appliances operating parameters

There is a wide difference in the energy efficiency of available fans in residential buildings and the super-efficient fans available in the market. The existing fan available at residential buildings consumes 75 W whereas a super-efficient fan consumes 35 W [78]. The fans have high usage in the residential building which is approximately around 1600 hours per year. The energy tariff for the residential consumer is in the range of Rs 7 to 8/kWh. The amount involved in the procurement of a star fan is recovered within 5.8 years. Similarly, the payback period for the replacement of a three-star 1.5-ton room air conditioner with an Energy Efficiency Ratio (EER) of 2.9 by a five-star energy label room air-conditioner with an EER of 5.0 is 8.5 years. The exercise of this replacement of the air conditioner involves an expenditure of Rs 25,000. The energy-saving and economics of the Energy Star label appliances are tabulated in Table 6.4 and Table 6.5 respectively.

| | | | | - |
|----------------------|----------|----------------------|--------------|----------|
| Residential | Quantity | Energy-saving | Total energy | Economic |
| Electric | | (kWh/appliance/year) | saving | saving |
| Appliance | | | (kWh/year) | [@ Rs |
| | | | | 7/kWh] |
| | | | | (Rs) |
| Room air conditioner | 02 | 852 | 1,704 | 11,928 |
| Ceiling Fan | 02 | 64 | 128 | 896 |
| Refrigerator | 01 | 345 | 345 | 2,415 |
| Total | | | 2,177 | 15,239 |

Table 6.4 Energy saving by replacement of residential building appliances.

| Payback | NPV (Rs) (d@9.36%) | CCE (Rs/ kWh) |
|-------------|---|--|
| period (yr) | | |
| 8.5 | 5,351.60 | 0.650 |
| | | |
| 5.8 | 3,069.44 | 3.96 |
| 9.2 | 4,054.35 | 5.51 |
| | Payback period (yr) 8.5 5.8 9.2 | Payback NPV (Rs) (d@9.36%) period (yr) |

Table 6.5 Economics of replacement of appliances

6.4 TECHNO-ECONOMIC ASSESSMENT OF ENERGY SAVING BY ROOFTOP SOLAR PV

Three rooftops (1kWp, 2kWp, and 3 kWp) solar PV systems are considered in this study. Energy generation by rooftop solar is evaluated by the RETScreen version 8. The software is open source and is available on the Natural Resources of Canada website [93]. For assessing the investment required for the installation of grid-integrated rooftop solar PV is taken from multiple suppliers, the same is tabulated in Table 6.6. The economics of the rooftop solar PV residential consumer is evaluated in terms of payback period and Net Present Value (NPV). For estimating the NPV, operation and maintenance cost (O&M) is considered 5% of the capital cost, During the life cycle of 25 years of rooftop solar PV a major recurring cost involved in the 13th year is the replacement of an inverter which is Rs 24,500, Rs 30,500 and Rs 37,500 for 1 kWp, 2 kWp and 3 kWp solar PV plant respectively. The economic parameters plugged into the RETscreen financial module are mentioned in Table 4.4. The results obtained from the RETscreen are tabulated in Table 6.7.

| Supplier | kWp (INR) | 2 kWp(INR) | 3 KWp (INR) |
|------------|-----------|------------|-------------|
| Supplier 1 | 64,636 | 1,02,872 | 1,43,308 |
| Supplier 2 | 70,000 | 1,40,000 | 1,95,000 |
| Supplier 3 | 76,000 | 1,55,000 | 2,25,000 |
| Average | 70,000 | 1,32,624 | 1,87,769 |

Table 6.6 Rooftop Solar Installation [79]

Table 6.7 Residential rooftop solar PV economics

| Plant | Electricity | Payback | NPV | CCE |
|--------------------|-------------|---------|----------|----------|
| Capacity | generation | period | (Rs) | (Rs/kWh) |
| (kW _p) | kWh/year | (Yr) | | |
| 1 | 1476 | 6.1 | 46,460 | 0.112 |
| 2 | 2952 | 4.7 | 1,56,500 | 0.112 |
| 3 | 4428 | 4.4 | 2,64,124 | 0.106 |

6.5. TECHO-ECONOMIC ASSESSMENT OF THE BUILDING ENVELOPE RETROFITMENT

The building under study is constructed using concrete and brick, the floor and the roof have a concrete structure; the floor surface is vinyl tiled whereas the walls are constructed using brick sand and cement with distemper available on the inner wall whereas the outer wall is plastered and not distempered but whitewashed. The building envelope (wall and roof) does not have any insulation. All the doors of the building are made of wood and the windows are standard single-glazed. The building block consists of four houses and each house is standard 3 BHK (Bedrooms, a Hall with a drawing room, and a Kitchen) with two washrooms. Requirement water pumping for the building is catered by centralized pumping which is not captured in the energy bill of the consumer. The Refrigeration and water heating requirement of the building is met by the installed electric geysers in the houses therefore these loads are clubbed with the plug load of the building. Space cooling is employed by the residential consumer from April to September for 9 hours/day. For these two rooms air conditioners of 2-ton capacity with an Energy Efficiency Ratio of 2.9 are deployed for the simulation. The occupancy of the house occupancy under study is considered four, a couple with two kids. To analyze the thermal characteristics of buildings and their impact on energy consumption eQUEST version 3.65.7175 software is used in this study. The software is user-friendly with the help of the building creation wizard tool of the software modeling the building can be carried out, and simulation results and energy consumption are graphically available for better appreciation. The eQUEST requires a weather file of the location where the building is located for the simulation, a compatible NewDelhi weather file from Energy Plus software is downloaded, and the same is used for the simulations. The impact on energy consumption of building envelopes i.e. wall and roof with and without insulation was analyzed using an EEM wizard tool available in the eQUEST software. For an economic analysis of the building envelope retrofitment, the cost of applying the insulation to the rooftop and walls is taken from the India Mart portal [94]. The building is simulated using the building wizard model and the eQUEST

building energy simulation results are shown in Figure 6.5 and Figure 6.6 respectively.



Figure 6.5 Building Model simulated by eQUEST



Figure 6.6 Distribution of the building's annual energy load

| Envelope | Insulation | Area | Energy | Economics |
|----------|----------------------|--------------------|--------------|-----------|
| | Cost | | conservation | [@ Rs |
| | (Rs/m ²) | | (kWh/year) | 7/kWh] |
| | | | | (Rs) |
| Roof | 1012 | 160 m ² | 2,000 | 14,000 |
| Wall | 733 | 240 m ² | 1,350 | 9450 |
| Total | | | 3,350 | 23,450 |

Table 6.8 Energy conservation by adding insulation to the building envelope

Table 6.9 Economics of Building Envelope Retrofit

| Envelope | Payback | NPV (Rs) | CCE (Rs/ kWh) |
|--------------|----------------|----------------------|---------------|
| (Insulation) | period (years) | (<u>d@9.36%</u> and | |
| | | n=30) | |
| | | | |
| Roof | 11.5 | -22,560 | 8.04 |
| | | | |
| Wall | 18.2 | -65,970 | 12.44 |
| | | | |

6.6 EPILOGUE

The additional EV load on the residential building will reduce the two energy stars considering the present energy consumption and prevailing energy star grading standard for the residential buildings. To compensate additional energy load of EVs, rooftop solar PV is the best choice among the options considered due to its favorable economics. Replacement of electrical appliances with the best energy efficiency appliance is also an option considered, the impact of this measure on the total energy consumption is not up to the desired level so it can mitigate the impact of EVs on the energy star of the building. Building envelope retrofit has the lowest economic returns among the options considered.

CHAPTER 7

CONCLUSION AND RECOMMENDATIONS FOR FUTURE WORK

7.1 CONCLUSION

The research aimed to take the stock of recently promulgated Energy Star label program for residential buildings and to address live issues of Energy Star labeling. The objectives of the study were to explore and suggest energysaving techniques for residential buildings so that the existing energy gap for five-star energy labels can be bridged. The sub-objectives of the study are to quantify the compromise on building energy stars by introducing EVs as the plugged load on the building. The different energy-saving measures along with the economic to mitigate the additional EV load are also captured in this study. Studies estimate that 2 kWp rooftops solar PV in the absence of state subsidy could not state could achieve socket parity in any Indian state. However, a 3 kWp rooftop solar PV plant could achieve socket parity in the absence of subsidy in five Indian states and if state subsidy is extended to consumers seven states could achieve socket parity. The study quantified the energy gap for an existing building in achieving a five-star energy label. The first step towards bridging this gap is to undertake an installation of rooftop solar PV considering the subsidy extended by the state. The study suggested the economics (payback period and NPV) of rooftop solar PV. The study also quantified the impact of EVs on the energy stars of the building and suggested

mitigation of this increased building load through energy-saving measures along with its economics.

By carrying out EV charging at home the building will have two reduced energy stars as compared to the building without EV. By installing 3 kW_p rooftop grid-integrated solar PV and replacing the old electrical appliances by 5-star energy appliances can mitigate the additional electrical load on the building and five-star label to the existing residential buildings. The study suggests that at grid-connected 3 kWp rooftops solar PV can bridge threefourths of the energy gap of the five energy stars for a building without an EV and half of the energy gap for a building with an EV as a plugged load. The invested amount for the installation of grid-connected rooftop solar PV systems can be recovered between 3 and 7 years. However, the payback period is more a function of state electricity tariff rather than solar radiation, The cities with the same solar radiations have different economies due to prevailing tariffs as each state has its energy tariff for residential consumption. For example, Pune and Ahmedabad cities receiving the same annual solar radiation have varied payback periods which are 3 years and 9 years respectively. By installing grid-intenerated rooftop solar PV, moderate energy consumers can bridge the energy gap for five-star energy labels. High-energy consumers need to take additional energy-saving measures like building envelope retrofitment and replacement of old room air conditioners with fivestar air conditioners for the star energy label of the building. The replacement of ceiling fans has the best economics among other electrical appliances. Insulation of the rooftop and wall has the worst economics among the options considered in this study, but it can cater to one-third of the energy gap for the five-star energy label.

7.2 FUTURE WORK

Star labeling for a residential building in India is presently voluntary and likely to be revised, as carried out for other star label programs under the gambit of the BEE star label program. Building technologies and energy-saving techniques are continuously evolving and the same be gelled with the revised Building Energy Star label which will be promulgated in the near future. The following work related to this study extension can be undertaken in the future: -

- (a) Exploring the Techno-economics analysis of Net zero residential building as BEE has recently published its grading in the form of "Sunya (Zero) and Sunya plus building".
- (b) Embodied energy has not been captured in India's Building Energy Star program and the same can be explored and suitable weightage can be awarded so that the life cycle energy consumption of the building is addressed holistically.
- (c) The development of a composite sustainable rating of the house needs to be explored which along with energy also encompasses water.

REFERENCES

- [1] "BP Report 2018" Retrieved 13 August 2024, from https://www.bp.com/content/dam/bp
- [2] S. Nalley and A. Larose, "IEO2021 Highlights," *Energy Inf. Adm.*, vol. 2021, p. 21, 2021, [Online]. Available: https://www.eia.gov/outlooks/ieo/pdf/IEO2021ReleasePresentation.pdf
- [3] "IEA statistics" IEA International Energy Agency. Retrieved 13 August 2022, from IEA website: https://www.iea.org/data-and-statistics
- [4] W. Amendments, ECBC 2017. Retrieved 10 August 2022, from https://beeindia.gov.in
- [5] E. Conservation, B. Code, and R. Buildings, *Eco-Niwas Samhita 2018*, vol. 1. 2018. Retrieved 13 August 2022, from https://beeindia.gov.in/content/ecbc-residential
- [6] Roberts, S. (2008). Altering existing buildings in the UK. *Energy* policy, 36(12), 4482-4486.
- [7] Chua, K. J., Chou, S. K., Yang, W. M., & Yan, J. (2013). Achieving better energy-efficient air conditioning–a review of technologies and strategies. *Applied energy*, 104, 87-104.
- [8] Zuo, J., & Zhao, Z. Y. (2014). Green building research–current status and future agenda: A review. *Renewable and sustainable energy reviews*, 30, 271-281.
- [9] American Council for an Energy-Efficient Economy, "The State Energy Efficiency Scorecard. Retrieved 12 August 2021, from http://aceee.org/state-policy/scorecardThe State Energy Efficiency Scorecard,"

- [10] Chan, A. T., & Yeung, V. C. (2005). Implementing building energy codes in Hong Kong: energy savings, environmental impacts and cost. *Energy and Buildings*, 37(6), 631-642.
- [11] Xu, L., Liu, J., Pei, J., & Han, X. (2013). Building energy saving potential in Hot Summer and Cold Winter (HSCW) Zone, China— Influence of building energy efficiency standards and implications. *Energy Policy*, 57, 253-262.
- [12] Jacobsen, G. D., & Kotchen, M. J. (2013). Are building codes effective at saving energy? Evidence from residential billing data in Florida. *Review of Economics and Statistics*, 95(1), 34-49.
- [13] Cheung, C. K., Fuller, R. J., & Luther, M. B. (2005). Energy-efficient envelope design for high-rise apartments. *Energy and buildings*, 37(1), 37-48.
- [14] Chan, K. T., & Chow, W. K. (1998). Energy impact of commercialbuilding envelopes in the sub-tropical climate. *Applied Energy*, 60(1), 21-39.
- [15] Balaras, C. A., Droutsa, K., Argiriou, A. A., & Asimakopoulos, D. N.
 (2000). Potential for energy conservation in apartment buildings. *Energy and buildings*, *31*(2), 143-154.
- [16] Tuominen, P., Klobut, K., Tolman, A., Adjei, A., & de Best-Waldhober,
 M. (2012). Energy savings potential in buildings and overcoming market barriers in member states of the European Union. *Energy and Buildings*, 51, 48-55.
- [17] IEA, I. (2013). Modernising Building Energy Codes to Secure our Global Energy Future. *International Energy Agency, Policy Pathway*, 8.
- [18] Qiu, Y., & Kahn, M. E. (2019). Impact of voluntary green certification on building energy performance. *Energy Economics*, 80, 461-475.

- [19] Li, H., & Carrión-Flores, C. E. (2017). An analysis of the ENERGY STAR® program in Alachua County, Florida. *Ecological Economics*, 131, 98-108.
- [20] Lee, S. E., & Rajagopalan, P. (2008). Building energy efficiency labeling programme in Singapore. *Energy Policy*, 36(10), 3982-3992.
- [21] Fuerst, F., & McAllister, P. (2011). Eco-labeling in commercial office markets: Do LEED and Energy Star offices obtain multiple premiums?. *Ecological economics*, 70(6), 1220-1230.
- [22] Elazab, R., Saif, O., Amin Metwally, A. M., & Daowd, M. (2022). New smart home energy management systems based on inclining block-rate pricing scheme. *Clean Energy*, 6(3), 503-511.
- [23] Gamero-Salinas, J., Monge-Barrio, A., Kishnani, N., López-Fidalgo, J.,
 & Sánchez-Ostiz, A. (2021). Passive cooling design strategies as adaptation measures for lowering the indoor overheating risk in tropical climates. *Energy and Buildings*, 252, 111417.
- [24] Prieto, A., Knaack, U., Auer, T., & Klein, T. (2018). Passive cooling & climate responsive façade design: Exploring the limits of passive cooling strategies to improve the performance of commercial buildings in warm climates. *Energy and Buildings*, 175, 30-47.
- [25] "Road map to fast track adoption and implementation of ECBC at urban and local level," 2020. Retrieved 12 August 2022, from http://www.aeee.in/wp-content/uploads/2018/11/AEEE-ECBC-Report.pdf
- [26] Tathagat, T., Prasad, A., & Nain, A. (2014). PACE-D technical assistance program HVAC market assessment and transformation approach for India. Technical Report, Nexant, Inc., USA, 2014. https://www.climatelinks.org/sites/default/files/asset/document/HVAC-Report-Send-for-Printing-Low-Res. pdf.

- [27] Dhaka, S., Mathur, J., & Garg, V. (2012). Combined effect of energy efficiency measures and thermal adaptation on air conditioned building in warm climatic conditions of India. *Energy and Buildings*, 55, 351-360.
- [28] Yu, S., Tan, Q., Evans, M., Kyle, P., Vu, L., & Patel, P. L. (2017). Improving building energy efficiency in India: State-level analysis of building energy efficiency policies. *Energy Policy*, 110, 331-341.
- [29] Manu, S., Wong, J., Rawal, R., Thomas, P. C., Kumar, S., & Deshmukh, A. (2011, November). An initial parametric evaluation of the impact of the energy conservation building code of India on commercial building sector. In Proc. Building Simulation 2011: 12th Conf. Int. Building Performance Simulation Assoc.(Sydney, 14–16 November) (pp. 1571-8).
- [30] "Report on Energy Efficiency and Energy Mix in the Indian Energy System (2030)," retrieved 13 August 2022, from http://niti.gov.in/mgov_file/Energy_Efficiency.pdf 2015.
- [31] Jain, M. A. K., Kumar, M. H., Sathis, M., Srinivas, S. N., Bhardwaj, M. A., Abdullah, M., ... & Kachhawa, M. S. (2017). Roadmap to fast track adoption and implementation of Energy Conservation Building Code (ECBC) at the urban and local level. *Conceived by: NITI Aayog I Supported by: BEE I Funded by: UNDP-GEF. Alliance for an Energy-Efficient Economy (AEEE)*, 3-4.
- [32] Khosla, R., Sagar, A., & Mathur, A. (2017). Deploying Low-carbon Technologies in Developing Countries: A view from India's buildings sector. *Environmental Policy and Governance*, 27(2), 149-162.
- [33] Mathur, A. (2019). Public costs and private benefits: The governance of energy efficiency in India. *Building Research & Information*, 47(1), 123-126.

- [34] Tulsyan, A., Dhaka, S., Mathur, J., & Yadav, J. V. (2013). Potential of energy savings through implementation of Energy Conservation Building Code in Jaipur city, India. *Energy and Buildings*, 58, 123-130.
- [35] Jayswal, M. (2012). To examine the energy conservation potential of passive & hybrid downdraught evaporative cooling: A study for commercial building sector in hot and dry climate of Ahmedabad. *Energy Procedia*, 30, 1131-1142.
- [36] Chedwal, R., Mathur, J., Agarwal, G. D., & Dhaka, S. (2015). Energy saving potential through Energy Conservation Building Code and advance energy efficiency measures in hotel buildings of Jaipur City, India. *Energy and Buildings*, 92, 282-295.
- [37] Chen, S. (2012). Civic agriculture: towards a local food web for sustainable urban development. *APCBEE Procedia*, *1*, 169-176.
- [38] Masoso, O. T., & Grobler, L. J. (2010). The dark side of occupants' behaviour on building energy use. *Energy and buildings*, 42(2), 173-177.
- [39] Busch, J. F. (1992). A tale of two populations: thermal comfort in airconditioned and naturally ventilated offices in Thailand. *Energy and buildings*, 18(3-4), 235-249.
- [40] Indraganti, M., Ooka, R., & Rijal, H. B. (2013). Thermal comfort in offices in summer: Findings from a field study under the 'setsuden'conditions in Tokyo, Japan. *Building and Environment*, 61, 114-132.
- [41] Hong, T. (2013). Occupant behavior: impact on energy use of private offices.
- [42] Hoyt, T., Arens, E., & Zhang, H. (2015). Extending air temperature setpoints: Simulated energy savings and design considerations for new and retrofit buildings. *Building and Environment*, 88, 89-96.

- [43] Praseeda, K. I., Mani, M., & Reddy, B. V. (2014). Assessing impact of material transition and thermal comfort models on embodied and operational energy in vernacular dwellings (India). *Energy Procedia*, 54, 342-351.
- [44] Rajasekar, E., & Ramachandraiah, A. (2010). Adaptive comfort and thermal expectations-a subjective evaluation in hot humid climate. *Proceedings of the adapting to change: new thinking on comfort. Windsor, London, UK*, 9-11.
- [45] Indraganti, M. (2010). Using the adaptive model of thermal comfort for obtaining indoor neutral temperature: Findings from a field study in Hyderabad, India. *Building and environment*, 45(3), 519-536.
- [46] Bódis, K., Kougias, I., Jäger-Waldau, A., Taylor, N., & Szabó, S. (2019). A high-resolution geospatial assessment of the rooftop solar photovoltaic potential in the European Union. *Renewable and Sustainable Energy Reviews*, 114, 109309.
- [47] Ordóñez, J., Jadraque, E., Alegre, J., & Martínez, G. (2010). Analysis of the photovoltaic solar energy capacity of residential rooftops in Andalusia (Spain). *Renewable and Sustainable Energy Reviews*, 14(7), 2122-2130.
- [48] Mansouri Kouhestani, F., Byrne, J., Johnson, D., Spencer, L., Hazendonk, P., & Brown, B. (2019). Evaluating solar energy technical and economic potential on rooftops in an urban setting: the city of Lethbridge, Canada. *International Journal of Energy and Environmental Engineering*, 10, 13-32.
- [49] Lee, M., Hong, T., Koo, C., & Kim, C. J. (2018). A break-even analysis and impact analysis of residential solar photovoltaic systems considering state solar incentives. *Technological and Economic Development of Economy*, 24(2), 358-382.

- [50] Dehwah, A. H., & Asif, M. (2019). Assessment of net energy contribution to buildings by rooftop photovoltaic systems in hot-humid climates. *Renewable energy*, 131, 1288-1299.
- [51] Hagerman, S., Jaramillo, P., & Morgan, M. G. (2016). Is rooftop solar PV at socket parity without subsidies?. *Energy Policy*, 89, 84-94.
- [52] Abul, S. B., Muhammad, E. H., Tabassum, M., Muscat, O., Molla, M. E., Ashraf, A., & Ahmed, J. (2020). Feasibility study of solar power system in residential area. *International Journal of Innovation in Computational Science and Engineering (IJICSE)*, 1(1), 10-17.
- [53] Gabr, A. Z., Helal, A. A., & Abbasy, N. H. (2020). Economic evaluation of rooftop grid-connected photovoltaic systems for residential building in Egypt. *International Transactions on Electrical Energy Systems*, 30(6), e12379.
- [54] Singh, R., & Banerjee, R. (2015). Estimation of rooftop solar photovoltaic potential of a city. *Solar Energy*, 115, 589-602.
- [55] India Solar Resource Data: Enhanced Data for Accelerated Deployment. (2016). In National Renewable Energy Laboratory. National Renewable Energy Laboratory. https://www.nrel.gov/docs/fy16osti/66070.pdf
- [56] Dondariya, C., Porwal, D., Awasthi, A., Shukla, A. K., Sudhakar, K., SR, M. M., & Bhimte, A. (2018). Performance simulation of gridconnected rooftop solar PV system for small households: A case study of Ujjain, India. *Energy Reports*, 4, 546-553.
- [57] Kotak, Y., Gago, E. J., Mohanty, P., & Muneer, T. (2014). Installation of roof-top solar PV modules and their impact on building cooling load. *Building Services Engineering Research and Technology*, 35(6), 613-633.

- [58] Shukla, A. K., Sudhakar, K., & Baredar, P. (2016). Design, simulation and economic analysis of standalone roof top solar PV system in India. *Solar Energy*, 136, 437-449.
- [59] Tomar, V., & Tiwari, G. N. (2017). Techno-economic evaluation of grid connected PV system for households with feed in tariff and time of day tariff regulation in New Delhi–A sustainable approach. *Renewable* and Sustainable Energy Reviews, 70, 822-835.
- [60] Hui, S. C. (2003, August). Effective use of building energy simulation for enhancing building energy codes. In Proc. of the IBPSA Building Simulation 2003 Conference (pp. 11-14)
- [61] O'Donnell, J., Keane, M., Morrissey, E., & Bazjanac, V. (2013). Scenario modelling: A holistic environmental and energy management method for building operation optimisation. *Energy and Buildings*, 62, 146-157.
- [62] Samuelson, H. W., Ghorayshi, A., & Reinhart, C. F. (2016). Analysis of a simplified calibration procedure for 18 design-phase building energy models. *Journal of Building Performance Simulation*, 9(1), 17-29.
- [63] O'Brien, W., Gaetani, I., Gilani, S., Carlucci, S., Hoes, P. J., & Hensen, J. (2017). International survey on current occupant modelling approaches in building performance simulation. *Journal of Building Performance Simulation*, 10(5-6), 653-671.
- [64] Gao, H., Koch, C., & Wu, Y. (2019). Building information modelling based building energy modelling: A review. *Applied energy*, 238, 320-343.
- [65] Horne, R., & Hayles, C. (2008). Towards global benchmarking for sustainable homes: an international comparison of the energy performance of housing. *Journal of Housing and the Built Environment*, 23, 119-130.

- [66] Yang, F., & Jiang, Z. (2019, September). Urban building energy modelling and urban design for sustainable neighbourhood development-A China perspective. In *IOP Conference Series: Earth* and Environmental Science (Vol. 329, No. 1, p. 012016). IOP Publishing.
- [67] Tsangrassoulis, A., Kontadakis, A., & Doulos, L. (2017). Assessing lighting energy saving potential from daylight harvesting in office buildings based on code compliance & simulation techniques: A comparison. *Procedia environmental sciences*, 38, 420-427..
- [68] Glasgo, B., Hendrickson, C., & Azevedo, I. L. (2017). Assessing the value of information in residential building simulation: Comparing simulated and actual building loads at the circuit level. *Applied Energy*, 203, 348-363..
- [69] Xie, Y., Mendon, V., Halverson, M., Bartlett, R., Hathaway, J., Chen, Y., ... & Liu, B. (2019). Assessing overall building energy performance of a large population of residential single-family homes using limited field data. *Journal of Building Performance Simulation*, 12(4), 480-493.
- [70] Hesaraki, B., Chen, Y., Dias Ferreira, R., & Al-Hussein, M. (2019).
 Energy code compliant house design for lowest lifecycle cost based on market-available technologies. *Canadian Journal of Civil Engineering*, 46(6), 308-321.
- [71] Bhatnagar, M., Mathur, J., & Garg, V. (2019). Development of reference building models for India. *Journal of Building Engineering*, 21, 267-277.
- [72] Farooq, J., & Sharma, P. (2018). A BIM-based Detailed Electrical Load Estimation, Costing and Code Checking. *International Journal of Electrical & Computer Engineering (2088-8708)*, 8(5).

- [73] Shivsharan, A. S., Vaidya, D. R., & Shinde, R. D. (2017). 3D Modeling and energy analysis of a residential building using BIM tools. *Int. Res. J. Eng. Tech*, 4(7), 629-636.
- [74] Ates, A. M., & Singh, H. (2021). Rooftop solar Photovoltaic (PV) plant–One year measured performance and simulations. *Journal of King Saud University-Science*, 33(3), 101361.
- [75] Thotakura, S., Kondamudi, S. C., Xavier, J. F., Quanjin, M., Reddy, G. R., Gangwar, P., & Davuluri, S. L. (2020). Operational performance of megawatt-scale grid integrated rooftop solar PV system in tropical wet and dry climates of India. *Case Studies in Thermal Engineering*, 18, 100602.
- [76] "Solar PV installation status." Retrieved 13 August 2024, from https://www.mnre.gov.in/solar/currentstatus/https://www.mnre.gov.in/solar/current-status/
- [77] "Rooftop solar PV installations." Retrieved 13 August 2024, from BRIDGE TO INDIA website:https://bridgetoindia.com
- [78] "Electricity tariff for residential consumer ." Retrieved 13 August 2024, from https://www.bijlibachao.com
- [79] "Rooftop solar PV prices." Retrieved 13 August 2024, from Kenbrook Solar website: https://kenbrooksolar.com/system/on-grid-solar-system.
- [80] "RE Tariff regulation in India Retrieved 13 August 2024, from http://www.cercind.gov.in/2020/draft_reg/DEM-RE-Tariff-Regulations2020
- [81] Mercom India clean energy news, insights, and analysis. (2016, June 23). Retrieved 13 August 2024, from Mercomindia.com website: https://mercomindia.com/

- [82] "Energy star labeling". Retrieved 12 August 2022, from https://beeindia.gov.in/sites/default/files/Labelling
- [83] "Energy Statistics India". Retrieved 12 August 2022, from https://mospi.gov.in/web/mospi/reports-publications
- [84] National portal for rooftop solar ministry of new and renewable energy. Retrieved 12 August 2021, from https://solarrooftop.gov.in/rooftop_calculator
- [85] International Energy Agency. (2022). Global EV Outlook 2022. doi:10.1787/c83f815c-en
- [86] Ahmed, A. A., Nazzal, M. A., Darras, B. M., & Deiab, I. M. (2022). A comprehensive sustainability assessment of battery electric vehicles, fuel cell electric vehicles, and internal combustion engine vehicles through a comparative circular economy assessment approach. *Sustainability*, 15(1), 171.
- [87] SMEV, Society Of Manufacturers Of Electric Vehicles. SMEV society of manufacturers of electric vehicles. Retrieved 13 August 2022, from https://www.smev.in/
- [88] Saxena, S., Gopal, A., & Phadke, A. (2014). Electrical consumption of two-, three-and four-wheel light-duty electric vehicles in India. *Applied energy*, 115, 582-590.
- [89] Nimesh, V., Sharma, D., Reddy, V. M., & Goswami, A. K. (2020). Implication viability assessment of shift to electric vehicles for present power generation scenario of India. *Energy*, 195, 116976.
- [90] Sharma, I., & Chandel, M. K. (2020). Will electric vehicles (EVs) be less polluting than conventional automobiles under Indian city conditions?. *Case Studies on Transport Policy*, 8(4), 1489-1503.

- [91] Schey, S., Scoffield, D., & Smart, J. (2012). A first look at the impact of electric vehicle charging on the electric grid in the EV project. *World Electric Vehicle Journal*, 5(3), 667-678.
- [92] Singh, V. K., Henriques, C. O., & Martins, A. G. (2018). Fostering investment on energy-efficient appliances in India–A multi-perspective economic input-output lifecycle assessment. *Energy*, 149, 1022-1035.
- [93] "Retscreen software." Retrieved 13 August 2020, from https://www.nrcan.gc.ca/maps-tools-publications/tools/data-analysissoftware-modelling/retscreen/
- [94] "Building retrofitment vendor ." Retrieved 13 August 2020, from https://www.indiamart.com
APPENDIX - 'A'

BIODATA

Wing commander Rakesh Dalal (Retd) with an experience of 23 years in the Indian Air Force in different technical assignments of expertise in planning, operations, project management, training & HR management, took premature retirement on 31 Dec 2022 and is presently employed as GM Defense solution with PCI gases India Pvt, Ltd.

RESEARCH PAPERS PUBLISHED

[1] R. Dalal, K. Bansal, and S. Thapar, "Bridging the energy gap of India's residential buildings by using rooftop solar PV systems for higher energy stars," *Clean Energy*, vol. 5, no. 3, pp. 423–432, 2021, doi: 10.1093/ce/zkab017.

[2] R. Dalal, K. Bansal, and S. Thapar, "Grid Parity of Residential Building Rooftop Solar PV in India," Strategic Planning for Energy and the Environment, Vol. 39 1–4, 19–40. doi: 10.13052/spee1048-4236.39142.

[3] R. Dalal and D Saini, "Mitigation of the impacts of electric vehicle charging on energy-star ratings for residential buildings in India. Clean Energy, 2023, Vol. 7, No. 5, 981–993, doi: https://doi.org/10.1093/ce/zkad041

[4] R. Dalal and D Saini, "Rooftop Solar PV Socket Parity for Commercial Buildings in India". 2023 5th International Conference on Energy, Power, and Environment: Towards Flexible Green Energy Technologies (ICEPE), doi.org/10.1109/ICEPE57949.2023.10201490

| RakeshJi_thesis_22Dec | | | | | |
|--|--|--|--|----------------------|--------|
| ORIGINALITY REPORT | | | | | |
| 8% SIMILARITY INDEX | | 3% INTERNET SOURCES | 7% PUBLICATIONS | 3% STUDENT | PAPERS |
| PRIMARY SOURCES | | | | | |
| Rakesh Dalal, Devender Kumar Saini. "Mitigation of the impacts of electric vehicle charging on energy-star ratings for residential buildings in India", Clean Energy, 2023 Publication | | | | | |
| 2 | Submitted to Indian Institute of Technology Jodhpur Student Paper | | | | 2% |
| 3 | academi Internet Source | c.oup.com | | | 2% |
| 4 | 4 Rakesh Dalal, Kamal Bansal, Sapan Thapar. "Bridging the energy gap of India's residential buildings by using rooftop solar PV systems for higher energy stars", Clean Energy, 2021 Publication | | | | <1% |
| 5 | dr.ddn.upes.ac.in:8080 | | | | <1% |
| 6 | Rakesh E "Rooftop Commer Internati | Dalal, Devender Solar PV Socke cial Buildings ir onal Conferenc | Kumar Saini et Parity for 1 India", 2023 e on Energy, | i. 3 5th Power | <1% |